UNIVERSITY of York

This is a repository copy of Housing Benefit exceptional hardship payments.

White Rose Research Online URL for this paper: <u>https://eprints.whiterose.ac.uk/id/eprint/73376/</u>

Monograph:

Sainsbury, R orcid.org/0000-0002-1217-6818 (1999) Housing Benefit exceptional hardship payments. Research Report. Department of Social Security Research Report, No. 91 . Corporate Document Services , Leeds.

Reuse

Items deposited in White Rose Research Online are protected by copyright, with all rights reserved unless indicated otherwise. They may be downloaded and/or printed for private study, or other acts as permitted by national copyright laws. The publisher or other rights holders may allow further reproduction and re-use of the full text version. This is indicated by the licence information on the White Rose Research Online record for the item.

Takedown

If you consider content in White Rose Research Online to be in breach of UK law, please notify us by emailing eprints@whiterose.ac.uk including the URL of the record and the reason for the withdrawal request.



eprints@whiterose.ac.uk https://eprints.whiterose.ac.uk/ DEPARTMENT OF SOCIAL SECURITY

HOUSING BENEFIT EXCEPTIONAL HARDSHIP PAYMENTS

by Roy Sainsbury



ę,

RESEARCH REPORT No

91

C

DEPARTMENT OF SOCIAL SECURITY

RESEARCH REPORT No 91

HOUSING BENEFIT EXCEPTIONAL HARDSHIP PAYMENTS: THE USE OF DISCRETIONARY POWERS

Roy Sainsbury

A report of research carried out by the Social Policy Research Unit at the University of York on behalf of the Department of Social Security

Corporate Document Services

© Crown Copyright 1999 Published for the Department of Social Security under licence from the Controller of Her Majesty's Stationary Office by Corporate Document Services, Leeds

Application for reproduction should be made in writing to The Copyright Unit, Her Majesty's Stationery Office St Clements House, 2-16 Colegate, Norwich NR3 1BQ

First Published 1999

ISBN 1 84123 104 5

Views expressed in this report are not necessarily those of the Department or any other Government Department

Printed by Thornton & Pearson (Printers) Limited

CONTENTS

_

| Acknowledgements vt |
|---|
| The Author vu |
| Executive summary 1 |
| 1 Introduction 7 Policy background 7 Anns of the research 7 Research design 8 Initial development stage 8 The telephone survey 9 Available data 10 Structure of the report 11 |
| 2 Expenditure on exceptional hardship payments 13 Introduction 13 Overall expenditure 13 Views about budgets 14 Discussion , 16 |
| 3 Applications for exceptional hardship payments 17 Introduction 17 Application rates 17 Levels of publicity and application rates 18 Potential demand 20 Discussion 22 |
| 4 Why clamants apply for exceptional hardship payments and how local authorities make decisions 23 Introduction 23 Why people apply for exceptional hardship payments 24 How local authorities make decisions on exceptional hardship payments 26 Relevance of pre-tenancy determinations in decision making 29 Content analysis of local authorities policies/criteria 29 Discussion 31 |
| 5 Outcomes of applications for exceptional hardship payments 33 Introduction 33 Successful applications and success rates 33 Successful applications 33 Success rates 34 Analysis of exceptional hardship payment recipients 36 Value of exceptional hardship payments 38 |

Щ

- -

Appeals 40 Discussion 42 6 Administering the scheme 45 Introduction 45 Views about administration 45 Discussion 46 7 Exceptional hardship payments - summary and discussion 49 The research questions 49 Comments on policy development 55 References 57 Appendix 1 - Questionnaire for telephone survey 59 Appendix 2 - Introductory letters to local authorities from DSS and SPRU 63

Other research reports available 69

LIST OF TABLES

| Table 1 1 | Sample of local authorities for the development stage 9 |
|-----------|---|
| Table 1 2 | The achieved sample by local authority type 9 |
| Table 1-3 | Analysis of non-response 10 |
| Table 2 1 | Expenditure on exceptional hardship payments 1997/98 14 |
| Table 2 2 | Views on local authorities' permitted totals, by level of |
| | expenditure on exceptional hardship payments 14 |
| Table 2.3 | Reasons why permitted totals were considered |
| | ʻabout r ight' 15 |
| Table 3-1 | Monthly application rates by local authority type 17 |
| Table 3 2 | Analysis of how local authorities publicise exceptional |
| | hardship payments 18 |
| Table 3 3 | Levels of publicity for exceptional hardship payments, by |
| | local authority type 19 |
| Table 3-4 | Levels of publicity and application rates 20 |
| Table 3 5 | Maximum quarterly rent restrictions as indicators of potential demand for exceptional hardship payments 21 |
| Table 4 1 | Claimants' reasons for requesting an exceptional hardship payment, distribution among local authorities 24 |
| Table 4 2 | Analysis of 'common' reasons for applications in local authorities with 20 or more applications, 1997/98 25 |
| Table 4 3 | Aids to decision making used by local authorities, by local authority type 27 |
| Table 4 4 | Approaches to decision making by local authority type 28 |

| Table 4 5 | Relevance of pre-tenancy determinations for deciding applications for exceptional hardship payments, by local authority type 29 |
|------------|---|
| Table 5-1 | Number of exceptional hardship payments awarded by local authority type 34 |
| Table 5 2 | Success rate of applications by local authority type 35 |
| Table 5-3 | Overall success rates compared with decision making approach of local authorities 36 |
| Table 5-4 | Success rates of different claimant groups 37 |
| Table 5 5 | Success rates of claimants with specific characteristics 37 |
| Table 5 6 | <i>Lowest</i> weekly exceptional hardship payments made by local authorities 38 |
| Table 5 7 | Average weekly exceptional hardship payments made by local authorities 39 |
| Table 5 8 | Estimate of average value of exceptional hardship payments to successful applicants 40 |
| Table 5.9 | Number of appeals, by local authority type 41 |
| Table 5-10 | Appeal rates within individual authorities with more than20 rejected applications42 |
| Table 6 1 | Experience of administering the exceptional hardship payment scheme, by local authority type 45 |
| Table 6 2 | Experience of administration, by rate of applications 46 |

ACKNOWLEDGEMENTS

The successful completion of this research project has been dependent on the goodwill and co-operation of Housing Benefit staff in over 300 local authorities in Great Britain who responded to the telehone survey My thanks to them all Particular thanks must go to the Housing Benefit Managers and staff in the 18 authorities whom we visited in the preliminary stages of the project. Their information, insights and suggestions were invaluable

The telephone survey was conducted on our behalf by Public Attitude Surveys Ltd of High Wycombe – My thinks for the haid work put into carrying out the interviews and chasing local authorities to Sandra Jowett, her predecessor Jude Cummins, and all the staff at PAS

Throughout the project we had regular contact with Rachel Trott of the DSS Social Research Branch and useful meetings with policy customers. Clive Singh and Jean Wavgood – Thank you to them for keeping the research on track and for their constructive input to all the stages of the project.

The project also relied heavily on my colleagues at the Social Policy Research Unit who carried out the research with their usual high level of expertise and enthusiasm. Thanks to Anne Corden, Julie Williams Jane Carlisle Jude England and Chris Jacobs for contributing to the fieldwork and to our backroom staff. Sally Pulleyn, Jenny Bowes, Teresa Frank and Lorna Foster, without whom nothing at all can happen.

Any responsibility for errors or omissions in the report remains with the Social Policy Research Unit, and the views expressed are not necessarily those of the DSS or other government department

THE AUTHOR

Dr Roy Samsbury is a Semor Research Fellow at the Social Policy Research Unit (SPRU) at the University of York where he has worked since 1988. His previous research has included projects on various aspects of the Housing Benefit system such as the review and appeals arrangements and recently, the Housing Benefit Verification Framework. His other work has included research on Income Support, the New Deal for Disabled People, disability benefits, and benefit fraud.

SPRU was established in 1973 as an independent research unit within the University's Department of Social Policy and Social Work employing around 25 research staff. The Unit has an international reputation for research in key areas of social policy including social security, and health and community care

EXECUTIVE SUMMARY

Introduction Local authorities have a range of discretionary powers which can be used in the administration of Housing Benefit. This study evaluates authorities use of discretionary powers in making exceptional hardship payments to claimants whose Housing Benefit does not cover the cost of their rent (under regulations in force since January 1996). The main objective of the research was to investigate how the system of exceptional hardship payments is operated by local authorities and, in particular, why expenditure has varied so significantly between authorities and, overall, has been lower than expected

The research methods comprised visits to 18 local authorities during which face-to-face interviews were conducted with Housing Benefit managers and assessment staff and a telephone survey of Housing Benefit managers in all authorities in Great Britain – Successful interviews were conducted in 305 local authorities, a response rate of 75 per cent

Expenditure on exceptional Each local authority has an allocation of funds from central government hardship payments for spending on exceptional hardship payments. An authority is also allowed to spend above this amount up to a ceiling, the 'permitted total

> In the first full year of the exceptional hardship payment scheme (1996/ 97) local authorities spent 27 per cent of the government allocation of \pounds 18 25 million. The results from the survey suggest that expenditure increased to around 48 per cent of the government allocation in 1997/ 98. The government funding for exceptional hardship payments was, therefore still under-spent to a considerable degree, and overall expenditure was well below the aggregate 'permitted total' of \pounds 42 million for all authorities

> Nearly a third of the authorities in the survey (30 per cent) spent less than ten per cent of their government allocation in 1997/98. A small number of authorities (18) reported spending nothing. All the authorities reporting expenditure in excess of their government allocation including the two highest spenders who had spent over twice the allocation were still within their permitted totals.

> Two out of three Housing Benefit managers thought that their allocation was about right', the most common reason being that it was expected that demand for exceptional hardship payments would increase in future

Applications for exceptional
hardship paymentsMost local authorities had to deal with very few applications for exceptional
hardship paymentshardship paymentshardship payments in 1997/98One in four authorities had fewer than
one application per month eight authorities reported receiving no
applications at all in the yearApplications at all in the yearApplication rates tended to be highest in
the London Boroughs and Metropolitan authorities

The research findings suggest that applications are increasing but are still at a low level. From the data on Rent Officer restrictions, there is seemingly a very large pool of potential applicants whose rent is not met in full by their Housing Benefit.

There is some evidence from the survey that publicising the availability of exceptional hardship payments in a range of different ways, rather than relying solely on decision letters, could increase applications. Many authorities were using combinations of sometimes innovative methods to inform claimants about exceptional hardship payments.

Local authorities can make an exceptional hardship payment in any case where it judges that the claimant or members of their family, would suffer exceptional hardship if the shortfall between the claimant's Housing Benefit and their rent is not met. There is no statutory definition of exceptional hardship' but the DSS has issued guidance to local authorities in Circular HB/CTB A7/96

The most common reasons put forward by claimants in their applications were as follows

- inability to afford rent,
- · existence of a medical condition illness or disability,
- mability to pay bills
- being under 25 years old (and therefore being subject to the 'single room rate'),
- need for room for children to stay,
- pregnancy,
- mability to pay for food,
- · need to get away from domestic violence

Most claimants included a number of reasons in their applications rather than a single one

Many authorities have developed their own approaches to deciding claims reflected in written policy documents, guidance to assessment staff, application forms, and publicity materials. Some authorities relied mainly on the DSS guidance. Others reported in the survey that they used neither the official guidance nor their own policy. Accordingly, local authorities' approaches to decision making could be characterised in the following ways

Why claimants apply for exceptional hardship payments and how local authorities make decisions

- the 'comprehensive' approach, using both DSS and own guidance (adopted by 50 per cent of authorities),
- the 'independent' approach, principally using own policies (19 per cent),
- the DSS-based' approach, principally using the official guidance (23 per cent),
- the 'case-based' approach, using neither form of guidance (nine per cent)

Using this typology the findings from the survey included the following

- Scottish authorities rely on DSS guidance the most, London Boroughs and Welsh authorities the least,
- · the independent approach is used most in London,
- more metropolitan authorities adopt the comprehensive approach than other authorities

The scope and content of the documentation used by local authorities in deciding applications varied enormously. Particularly striking was the number of authorities who require applicants to complete detailed expenditure forms as part of their application. Some authorities had devised scoring systems for assessing applications.

Housing Benefit claimants can ask a local authority to supply them with a *pre-tenancy determination* giving an indication of the likely amount of benefit they would receive if they took on a particular tenancy. The large majority of authorities take pre-tenancy determinations into account when deciding applications for exceptional hardship payments, although the existence of a determination would not be a sufficient reason to reject an application

Outcomes of applications for exceptional hardship payments Of the 305 local authorities in the survey, 288 authorities supplied figures for the number of exceptional hardship payments awarded in 1997/98 Of these, 248 were also able to supply figures for the number of unsuccessful applications Aggregating the responses from these latter authorities shows that 9,483 exceptional hardship payments were made from 22,034 applications, an overall success rate of 43 per cent. The mean number of payments in these authorities was 38 within a range of between one and 664

> One in five authorities were making some relatively high weekly payments, in excess of $\pounds 50$ and a small number had made payments of over $\pounds 100$ per week. However almost all authorities reported *average* weekly payments of $\pounds 30$ or less. Exceptional hardship payment cases then tended to fall into one of two types cases where the nature of the exceptional hardship was considered temporary, which would be removed when the claimant found more suitable accommodation, and cases where the nature of the exceptional hardship was considered permanent and which could not be alleviated by a move to other accommodation

> > <u>3</u>

| , | The survey data suggest that people whose circumstances are related in some way to their health (disabled people those with mental health problems, or pregnant women) are the most successful types of claimant However, the most numerous beneficiaries of exceptional hardship payments were lone parents. Pensioners appeared to fare badly in the sense that applications were seemingly low compared with the large number of pensioners in the population. |
|---|---|
| | Wide variations between local authorities were also apparent in the range of exceptional hardship payments made and the average amounts paid to individual claimants. In some authorities apparently large weekly payments and large aggregate amounts were the norm rather than the exception |
| | Relatively few unsuccessful applicants pursued their cases to appeal (or internal review in the first instance). The success rate of about one in four is in line with success rates for other means-tested benefits |
| Administering the scheme | Just under a half of the authorities in the survey had experienced some difficulties in the administration of exceptional hardship payments and a small number (eight authorities) reported that they had had 'serious problems'. The most common difficulties were |
| | lack of guidance on what constitutes exceptional hardship (mentioned by 51 local authorities), computer problems including setting up systems and with software (43), |
| | ensuring staff were sufficiently trained and equipped (22) length of time needed to investigate applications (20), keeping track of cases and expenditure (18), amount of work involved gathering information (17) |
| | There was no consistent pattern in the types of difficulties reported by the eight local authorities with serious problems. Only one of the eight said that they had had computer-related problems |
| Exceptional hardship payments - Summary and Discussion | Variations between local authorities in the wavs they administer aspects of Housing Benefit policy are the norm rather than the exception. It is not surprising, therefore, to find wide variations in authorities administration of the exceptional hardship payment scheme |
| | On the basis of this research it is fair to infer that applicants for exceptional hardship payments and importantly potential applicants receive very different treatment in different local authority areas. Local variations in rent levels, the number and amounts of rent restrictions and the demographic constitution of the claimant population will all contribute to differences in application rates, success rates and value of exceptional hardship payments. However, these variations cannot explain why some authorities have zero application rites and zero success rates while other authorities attract hundreds of applications, and some have 100 pet cent success rates. |

success rates

There is some, though not conclusive, evidence from the study that the low level of applications and awards suggest that the guidance set out in Circular A7/96 could possibly be drawn too tightly such that local authorities are making decisions that exclude claimants who are suffering a degree of considerable hardship but not to the extent suggested in the guidance. The policy of exceptional hardship payments may therefore, not be helping some of the people for whom it was intended

The variations discovered in the documents supplied by local authorities in the course of the research suggest that there is a case for a good practice' guide that contains examples of high quality publicity materials, appropriate wording to be used in decision letters application forms, and internal documents used in decision making

The policy option of reducing the budget for exceptional hardship payments is always available. The research suggests that the demands on the budget are increasing and likely to continue increasing. Housing Benefit managers are aware of this also. Any proposed reduction in budgets is therefore likely to generate opposition from a large number of authorities.

ž

I INTRODUCTION

In January 1998 the Department of Social Security commissioned the Social Policy Research Unit to carry out research into local authorities' use of new discretionary powers to make exceptional hardship payments to claimants whose Housing Benefit does not cover the cost of their rent The fieldwork for the project was carried out in February to June 1998

Since January 1996, for most new claims for Housing Benefit from tenants Policy background in the private deregulated sector, calculations of awards have not been based on the actual rent paid by the claimant but on a local authority's 'local reference rent' a figure that reflects the general level of rents for properties of a similar size in the locality In October 1996, similar restrictions were placed on single claimants under the age of 25 making new or renewal claims by using a local 'single room rent' for calculating Housing Benefit awards rather than the local reference tent. The single room rent is a figure reflecting the average cost of non self-contained accommodation without board in the locality Both changes were intended to encourage claimants to seek 'reasonably priced' and appropriate accommodation. However, it has long been recognised that, in exceptional circumstances, the welfare of individual claimants and their families requires that a higher amount of Housing Benefit be paid than that allowed under a strict application of the regulations Therefore, for many years, local authorities have had powers to vary the amount of an award m such circumstances In January 1996, when local reference rents were introduced, new powers were introduced enabling local authorities to pay an additional amount to bring Housing Benefit up to the level of the contractual rent (less meligible charges) where, in the assessment of the authority, the claimant or a family member would otherwise suffer exceptional hardship In October 1996 the scheme was extended to cover people affected by the introduction of the single room rent' regulations Exceptional hardship is not defined in legislation but local authorities have guidance on its interpretation in a DSS Housing Benefit Circular, A7/ 96

In 1996/97 expenditure on exceptional hardship payments was under $\pounds 5$ nullion compared with the ceiling on expenditure (the 'permitted total') of around $\pounds 42$ nullion (to which the central government contribution was $\pounds 18.25$ million)

Anns of the research The main objective of the research was to investigate how the system of exceptional hardship payments is operated by local authorities and, in particular, why expenditure has varied so significantly between authorities and overall, has been lower than expected

The specific aims of the project, set out below, were agreed between DSS and SPRU on the basis of the DSS's original research specification and SPRU's research proposal. Briefly the aims were to investigate the following research questions

- Why do local authorities spend very different amounts on exceptional hardship payments?
- Why is expenditure lower than expected?
- What are local authorities views about their allocations for exceptional hardship payments from central government?
- What are the patterns of applications and awards for exceptional hardship payments between local authorities?
- How do claimants become aware of the availability of exceptional hardship payments?
- Do particular types of claimants tend to be more or less successful in applying for an exceptional hardship payment than others?
- · How many applications are made, and how many are successful?
- How much is the average exceptional hardship payment?
- How do local authorities make decisions on applications for an exceptional hardship payment?
- What criteria do they use?
- Does a pre-tenancy determination affect the decision whether to make a payment?

Т

- How easy or difficult has it been to implement the exceptional hardship payment scheme?
- What is the nature of any problems experienced?
- Research design The main element of the research design was a telephone survey of all local authorities in England, Scotland and Wales This was conducted on behalf of SPRU by Public Attitude Surveys Ltd (PAS) of High Wycombe The survey was preceded by a development stage comprising visits to 18 local authorities
- *linutal development stage* The principal objective of the development stage of the project was to collect information on the main research issues in order to inform the development of a questionnaire for use in the telephone survey. The 18 local authorities were selected on the basis of the following criteria
 - size of Housing Benefit workload (using benefit expenditure as an indicator),
 - type of authority,
 - geographical area,
 - expenditure on exceptional hardship payments (using DSS data for the first half of 1997/98)

Table 1.1 Sample of local authorities for the development stage

| Number |
|--------|
| 8 |
| 3 |
| 3 |
| 2 |
| 2 |
| 8 |
| |

In each authority we conducted interviews with a Housing Benefit Manager and one or more officers with responsibility for making exceptional hardship payment decisions - Fieldwork visits took place in late February and early March 1998

The telephone survey A pilot of the telephone interviews was conducted with six local authorities A copy of the final questionnaire can be found in Appendix 1 Housing Benefit Managers in the 412 local authorities in Great Britain were sent introductory letters from DSS and SPRU (reproduced as Appendix 2) a background document explaining the research and a copy of the questionnaire¹ PAS conducted interviews in April and May Successful interviews were carried out with 305 managers a response rate of 75 per cent Table 1.2 shows the response rate for each local authority type

| Table 1.2 | The achieved sample by local authority type | |
|-----------|---|--|
| | | |

| | Possible | Achieved | Response rate (%) |
|--|----------|----------|-------------------|
| English District and Unitary authorities | 289 | 221 | 76 |
| English Metropolitan authorities | 36 | 27 | 75 |
| London authorities | 33 | 19 | 58 |
| Welsh authorities | 22 | 15 | 68 |
| Scottish authorities | 32 | 23 | 72 |
| Total | 412 | 305 | 75 |

The table shows that London authorities (i.e. the 32 London Boroughs and the City of London) are under-represented in the achieved sample Since these authorities are generally larger than most other authorities their under-representation affects the overall representativeness of the achieved sample From data supplied by DSS we compared the nonresponse authorities with the total population of authorities by the size of

During the fieldwork information was also collected on local authorities use of a separate set of new discretionary powers aimed at increasing the scope of counterfraud activity. The results of this research appear in the DSS' in-house research report series

their privately rented sector (using rent allowance expenditure as an indicator) and their expenditure on exceptional hardship payments (using data for the first half of 1997/98). Table 1-3 presents the results of the comparison

Table 1.3 Analysis of non-response

| Percentage of authorities in each | Non-response | All local authorities |
|--------------------------------------|--------------|-----------------------|
| quartile of the rent allowance | authorities | |
| distribution | | |
| l" quartile | 29 0 | 25.5 |
| 2 [™] quartile | 247 | 25 3 |
| 3 rd quartile | 19.4 | 242 |
| 4 th quartile | 269 | 25 0 |
| Percentage of authorities in each | | |
| quartile of the exceptional hardship | | |
| payment distribution | | |
| I" quartile | 20 2 | 249 |
| 2™ quartile | 25 5 | 249 |
| 3 rd quartile | 24 5 | 25 2 |
| 4 th quartile | 29.8 | 24 9 |
| Base | 94 | 380 |

Source Department of Social Security

The table shows that, as suggested above, we have lost from our achieved sample more of the larger authorities than expected – We have also lost more of the smaller authorities (by size of private sector) – From previous research work with local authorities this outcome is not surprising. Many small authorities appear to be reluctant to engage with research projects sometimes because their resources are small, but also because they do not see the relevance of the research to their authority – However, the two most common reasons cited by Housing Benefit managers from nonparticipating authorities were inability to meet the tight deadlines for the project, and the amount of work that would have been needed to gather the information we required

Available data As well as the data generated from the survey the following data were also made available by the DSS at the local authority level

- Expenditure on Housing Benefit for 1995/96
- Expenditure on rent allowances for 1995/96
- Allocations to local authorities for exceptional hardship payments for 1997/98,
- Expenditure on exceptional hardship payments in first half year of 1997/98,
- Restricted rent statistics for each local authority for the first three quarters of 1997/98

As mentioned earlier, the rent allowance and exceptional hardship payment expenditure data have been used to group local authorities into quartiles for the purposes of comparing the non-response authorities with all local authorities. The half year exceptional hardship payment expenditure data were used in the selection of local authorities for the development stage and to calculate expenditure rates for each authority. The rent restriction data were used as proxy indicators of the potential demand for exceptional hardship payments in each authority.

The final source of data was a selection of written policy documents and list of criteria that local authorities use to help them in making decisions on individual applications for exceptional hardship payments. These were requested from 161 authorities, of whom 88 responded. Their contents are analysed in Chapter 4.

Chapter 2 analyses local authorities' expenditure on exceptional hardship Structure of the report payments using official DSS statistics and data from the survey. Housing Benefit managers' views on the allocation of funds and the adequacy of their allocations are also explored. In Chapter 3 the survey data on application rates are analysed and compared with local authorities' approaches to publicising the exceptional hardship payment scheme. The potential demand and take up for exceptional hardship payments are also examined Chapter 4 presents the data on why people apply for exceptional hardship payments, and contains an analysis of how local authorities make decisions on applications, including their use of the DSS guidance and their own local policies Chapter 5 looks at the outcomes of applications, including an analysis of the amounts awarded and success and failure rates by claimant types. The chapter also examines the survey data on appeals against adverse decisions by claimants. Chapter 6 turns its attention to the administration of the scheme, and in particular identifies the types of problems encountered by local authorities in implementing the new regulations. Chapter 7 presents a summary of the findings on exceptional hardship payments and a discussion of their implications for policy development

<u>11</u>

2 EXPENDITURE ON EXCEPTIONAL HARDSHIP PAYMENTS

- Introduction This chapter analyses the expenditure of local authorities on exceptional hardship payments in the year 1997/98, using the survey data and compares it with the overall budget for the scheme. Housing Benefit managers' views on the adequacy of their budgets are also examined.
- Overall expenditure The amount of money that can be spent on exceptional hardship payments is cash-limited When the scheme was introduced in January 1996, the ceiling for expenditure was set at 0.9 per cent of in-year expenditure on deregulated tenancies. In October 1996 this percentage was increased to 1.08 per cent, yielding a permitted total of around \pounds 42 million for the year 1996/97. The central contribution to this budget was \pounds 18.25 million The intention was that expenditure above that amount would be met from local authorities own budgets. Actual expenditure for the first full year of the scheme was actually less than \pounds 5 million. This represents around 27 per cent of the government contribution and less than 12 per cent of the permitted expenditure

For 1997/98 the budget constraints were the same as for the first year. The central government contribution was frozen at \pounds 18 25 million with the permitted total remaining at 1.08 per cent giving an expenditure ceiling of \pounds 47 million. At the time the fieldwork was carried out the DSS had not vet collated details of expenditure on exceptional hardship payments for 1997/98. Hence, in the survey for this project local authorities were asked to provide as accurate a figure as possible for exceptional hardship payment expenditure for the period 1 April 1997 to 31 March 1998. Information was provided by 295 of the 305 local authorities in the survey and showed that these authorities spent nearly \pounds 6 million in 1997/98, representing around 48 per cent of their government contribution

Table 2.1 presents an analysis of local authorities spending as a percentage of their government allocation broken down by local authority type Spending above 100 per cent is legitimate but must be financed from the local authority's own budget provided it does not exceed the permitted total for that authority

| Type of authority | Per | Percentage of government allocation spent | | | | | | Base | |
|----------------------|-----|---|-----|------|-----|------|--------|------|-----|
| | | (% of authorities) | | | | | | | |
| . | 0- | 10 | | -20 | 2 | -50 | 51 and | over | |
| | No | (%) | No. | (%) | No. | (%) | No | (%) | |
| English District and | | (31) | 45 | (21) | 64 | (30) | 38 | (18) | 213 |
| Unitary authorities | | | | | | | | | |
| English Metropolitan | 5 | (19) | 4 | (15) | ŀΟ | (37) | 8 | (30) | 27 |
| authorities | | | | | | | | | |
| London authorities | 6 | (33) | 3 | (17) | 3 | (17) | 6 | (33) | 18 |
| Welsh authorities | 5 | (36) | 3 | (21) | 4 | (29) | 2 | (14) | 14 |
| Scottish authorities | 8 | (35) | 5 | (22) | 7 | (30) | 3 | (13) | 23 |
| All authorities | | (30) | | (20) | | (30) | | (19) | 295 |

Table 2.1Expenditure on exceptional hardship payments1997/98

Table 2.1 shows that nearly a third of the authorities in the survey (30 per cent) spent less than ten per cent of their government allocation. A small number of authorities (19) reported spending nothing. These were mostly small District or Scottish authorities. All the authorities reporting expenditure in excess of their government allocation, including the two highest spenders who had spent over twice the allocation, were still within their permitted totals. The numbers of London and Welsh authorities in the sample were small (18 and 14 respectively) so percentages based on them must be treated with caution. However, there is an indication from the table that London authorities and English Metropolitan authorities tend to spend larger proportions of their allocations than other authorities.

Views about budgets In the survey, Housing Benefit managers were asked their views about the amount they were allowed to spend on exceptional hardship payments (i e their 'permitted total rather than their government allocation) Table 2.2 presents their responses analysed by their level of expenditure

Table 2.2Views on local authorities' permitted totals, bylevel of expenditure on exceptional hardship payments

| Level of expenditure (% of government | Views of Housing Benefit managers about permitted total (percentage of local authorities) | | | | |
|--|---|------------|-------------|-----|--|
| | | | | | |
| | Too much | Too little | About right | | |
| 0-10 | 46 | 2 | 51 | 84 | |
| -20 | 34 | 5 | 61 | 56 | |
| 21-50 | 22 | 7 | 71 | 86 | |
| 51 and over | 11 | 9 | 80 | 56 | |
| All authorities | 29 | 6 | 65 | 282 | |

The table shows, as might be expected, that local authorities with low levels of exceptional hardship payment expenditure were more likely to consider that their permitted total was too much than higher spending authorities among which there was a greater tendency to consider their budgets about right'

Given the number of authorities which could be considered to be 'low spending', it is surprising that almost two-thirds thought their permitted total was about right. In order to explore the basis for the views expressed Housing Benefit managers were asked to give reasons for their answers

Of the 29 Housing Benefit managers who said that their permitted total was too much, the majority mentioned all of the following reasons

- Applying the DSS guidance meant that very few awards could be made
- The budget was too large in relation to successful claims
- There was a lack of demand for exceptional hardship payments
- There was a lack of awareness among claimants about the availability of exceptional hardship payments

The first two of these reasons are similar and make the point that the criteria for making exceptional hardship payments are (or are perceived to be) strict, such that too few claimants qualify to justify the size of their budgets

Table 2.3 presents an analysis of the reasons given for the permitted total being about right

| Table 2 3 | Reasons why permitted totals were considered | |
|-------------|--|--|
| 'about righ | ť | |

| Reason | Number of Housing Benefit | % |
|----------------------------------|----------------------------|----|
| | managers mentioning reason | |
| Expectation that demand for | 77 | 42 |
| exceptional hardship payments | | |
| will increase in future | | |
| Because the local authority | 45 | 25 |
| has not exceeded its budget | | |
| The budget allows a certain | 36 | 20 |
| amount of flexibility | | |
| Because supply is matching demar | nd 19 | 10 |

The most common reason for thinking the budget for exceptional hardship payments was about right, even though current expenditure was low, was that the demand for exceptional hardship payments was expected to rise in future. This comment is perhaps more a reflection of some managers' desire not to see the budget reduced rather than an endorsement of current levels. The general picture that seems to emerge from these comments and from analysis of the visits to local authorities in the development stage is that current levels of allowable expenditure are generally considered acceptable because they

- are greater than actual expenditure,
- · are adequate to cope with increases in demand if they happen
- allow local authorities to implement the exceptional hordship payment in their own way

Local authorities seemed generally to feel that at the current level of demand exceptional hardship payment budgets are not under stress

Discussion In the first year of the exceptional hardship payment scheme local authorities spent 27 per cent of the government allocation to the cashlimited budget of \pounds 42 million. The results from the survey suggest that expenditure has increased to around 48 per cent. The budget is, therefore still under-spent to a considerable degree. However, it is to be expected that expenditure rose in the second full year (1997/98), particularly since the scope of the scheme was extended in October 1996 to cover 'single room rent' claimants.

Low expenditure in the first year is also to be expected. Local authorities will always implement new provisions and schemes at different rates. Some authorities are more prepared or have a greater need than others to implement measures quickly. Furthermore, it is to be expected that knowledge of the scheme among claimants, and experience of how to succeed in getting a payment, will increase in the future leading to higher rates of applications and awards.

3 APPLICATIONS FOR EXCEPTIONAL HARDSHIP PAYMENTS

- Introduction This chapter examines the survey data on applications for exceptional hardship payments. Comparisons are made between application rates and the levels of publicity used by local authorities about exceptional hardship payments and with an indicator of the potential demand for exceptional payments derived from DSS data on Rent Officer restrictions
- Application rates The number of applications for exceptional hardship payments for each authority in the survey was calculated by adding the number of awards to the number of refusals. Not all authorities were able to give a figure for the number of refusals because records of refusals are not routinely kept. For the same reason some other authorities were able to supply estimates only. The number of applications, based on the sum of the figures supplied for successful and unsuccessful applications is therefore also an estimate. Table 3-1 presents an analysis of application rates by type of nuthority for the 267 authorities which provided data.

| Type of authority | Number of applications per month | | | | | | Base | | |
|----------------------|----------------------------------|------|----|------|----|------|------|------|-----|
| | | | > | -5 | >! | 5-10 | >10 | | |
| | No | (%) | No | (%) | No | (%) | No | (%) | |
| English District and | 59 | (30) | 89 | (45) | 24 | (12) | 27 | (14) | 199 |
| Unitary authorities | | | | | | | | | |
| English Metropolitan | 0 | (0) | 3 | (17) | Т | (o) | 14 | (78) | 18 |
| authorities | | | | | | | | | |
| London authorities | I | (7) | 2 | (14) | Ι | (7) | 10 | (71) | 14 |
| Welsh authorities | 1 | (8) | 9 | (59) | 1 | (8) | 2 | (15) | 13 |
| Scottish authorities | 10 | (45) | 10 | (45) | Ι | (5) | i | (5) | 22 |
| All authorities | | (27) | | (43) | | (10) | | (20) | 267 |

Table 3.1Monthly application rates, by local authority type(number of authorities, with row percentages in brackets)

The table shows that most authorities hild to deal with few applications for exceptional hardship payments in 1997/98. The lowest category (fewer than one application a month) includes eight authorities who reported no applications at all in the year. These were mostly small District authorities. As might be expected, application rates tended to be highest in the London and Metropolitan authorities which have larger Housing Benefit caseloads than most other authorities. Application rates could be related to a number of factors, two of which the survey and DSS data allow us to explore These are

- the levels of publicity given to the exceptional hardship payment scheme
- the potential demand for applications
- Levels of publicity and In the development stage of the research project, local authorities were asked how they publicised the availability of exceptional hardship payments. The most common responses were (a) decision letters notifying claimants that their award of Housing Benefit did not cover the full rent because it had been restricted by the Rent Officer. (b) information leaflets, (c) some form of special application form or tear-off slip on letters, and (d) in information supplied to local advice agencies. In the survey, therefore, local authorities were asked if they used any of these forms of publicity, and asked to specify any other publicity they used. Table 3.2 presents local authorities' responses.

Table 3.2Analysis of how local authorities publiciseexceptional hardship payments

| Type of publicity | Number of authorities | % (of 305 | |
|-----------------------------|------------------------------|--------------|--|
| | using this form of publicity | authorities) | |
| In decision letters | 237 | 78 | |
| Leaflets | 188 | 62 | |
| Separate form/tear off slip | 02 | 33 | |
| In information provided to | 206 | 68 | |
| advice agencies | | | |
| No publicity at all | 10 | 3 | |

There were a range of other methods of publicity mentioned in the survey responses including

- Posters in public places (mentioned by 20 authorities)
- Meetings with landlords (15)
- Newsletters or local press (13)
- Special information packs (9)
- Informing claimants in contact with local authority departments (7)
- Meetings with welfare groups/tenants associations (4)
- Direct mailshots (3)

It is perhaps surprising that 22 per cent of the sample (68 authorities) said they did not publicise exceptional hardship payments in the decision letters sent to claimants. Of these, ten authorities reported that the scheme was not publicised in any way in their authorities (five of which were small Scottish authorities).

Trying to isolate the impact of publicity on application levels is problematic, particularly because as we explain in the next section we only have rough indicators of some of the other factors which might have an influence, such as the number of claimants whose rent is restricted Furthermore we have no information directly from claimants about the role that publicity might have played in their decisions about applying for an exceptional hardship payment. Nevertheless we can look at the relationship between different levels of publicity and applications. The intention is not to establish causal relationships but to draw lessons for the development of policy and practice. Table 3.3 begins by comparing levels of publicity with local authority type.

Table 3.3 Level of publicity for exceptional hardship payments, by local authority type (number of authorities, with row percentages in brackets)

| Authority type | | | Leve | el of pu | iblicity | 7 | | | | | Base |
|-------------------------------------|-----|------|------|----------|-----------------|-------------------------|----------------------------------|--------------------|-----|----------------------------|------|
| | N | one | | tter | pl or oth | iter us ne ner | Let plus mo oth meth | 2 or ore ier | but | letter other olicity | |
| | No. | (%) | No | (%) | No | (%) | No | (%) | No | (%) | |
| English District and | 3 | (1) | 21 | (10) | 42 | (19) | 119 | (54) | 36 | (16) | 221 |
| Unitary authorities | | | | | | | | | | | |
| English Metropolitan authorities | 0 | (0) | 1 | (4) | 5 | (19) | (8 | (07) | 3 | (11) | 27 |
| London authorities | l | (5) | 0 | (0) | 2 | (1) | 11 | (58) | 5 | (26) | 19 |
| Welsh authorities | 1 | (7) | ł. | (7) | 3 | (20) | 4 | (27) | 8 | (40) | 15 |
| Scottish authorities | 5 | (22) | 3 | (13) | 2 | (9) | 5 | (22) | 8 | (35) | 23 |
| All authorities | | (3) | | (9) | | (18) | | (52) | | (19) | 305 |

Including information about exceptional hardship payments in decision letters is clearly the simplest way of telling claimants about their availability However, there is research evidence that some social security claimants either do not or cannot read or understand official letters (Stafford et al , 1997) Presumably with this in nund, the large majority of those authorities who do send information in letters also use some other form of publicity, the most popular being dissemination of information through local advice agencies. We do not know from the survey data why nearly one in five authorities have chosen an alternative way (or ways) of publicising exceptional hardship payments to letters, but Table 3.3 suggests that Welsh and Scottish authorities in particular are more likely to find alternatives than English authorities

The effect of publicity will depend on its quality as well as its quantity We know from the development stage, for example, that there are variations in the amount of information provided in letters and in the tone and wording used. Some authorities provided a very brief statement and invited claimants to contact the local authority, while others went into more detail. Some Housing Benefit managers expressed a concern to balance the need to inform claimants sufficiently without generating excessive demands or raising false expectations

Unlike decision letters other forms of publicity (with the possible exception of direct mailshots) are not guaranteed to reach the population of potential applicants. However, we cannot conclude from this that authorities relying on alternatives to letters are taking a restrictive approach to publicity. For example, our analysis shows that over half of these authorities publicised exceptional hardship payments in at least two other ways most often through the supply of information to local advice agencies. Although the numbers of applications depends on many factors, a comparison of applications and levels of publicity does yield some useful insights.

| Levels of publicity | Number | of applicati | ons per moi | nth | Base | |
|---------------------|--------------------------|--------------|-------------|-----|------|--|
| | (% of local authorities) | | | | | |
| | < | >1-5 | >5-10 | >10 | | |
| Letter only | 48 | 48 | 4 | 0 | 23 | |
| Letter plus one | 36 | 39 | ET. | 14 | 44 | |
| other method | | | | | | |
| Letter plus two | 17 | 44 | 11 | 29 | 139 | |
| other methods | | | | | | |
| No letter but other | 33 | 43 | 12 | 12 | 51 | |
| publicity | | | _ | | | |
| All authorities | 27 | 43 | 10 | 20 | 266 | |

| Table 3.4 Levels of publicity and application rate | Table 3 4 | Levels of publicity | and ap | plication | rates |
|--|-----------|---------------------|--------|-----------|-------|
|--|-----------|---------------------|--------|-----------|-------|

The pattern of applications for those authorities using decision letters suggests strongly that additional publicity does encourage more claimants to ask for exceptional hardship payments. The pattern for the authorities which use alternatives to letters is less easy to interpret, but the evidence for example that 12 per cent of these (i.e. six authorities in the survey) still attracted more than ten applications a month possibly lends support to the suggestion earlier that the *quality* of information has an important influence on application levels.

Potential demand The other factor that might influence application levels that we are able to explore is the number of people having their rent restricted. Every Housing Benefit claimant whose award is based on a restricted rent, using either a local reference tent or a single room rate as entitled to make an application for an exceptional hardship payment. These claimants represent the potential demand for exceptional hardship payments. From the visits to local authorities in the development stage it became clear that although there was no practical way for Housing Benefit departments to measure potential demand, there may be a large unmet demand for exceptional hardship payments among the claimant population. The survey responses reported in Chapter 2 present a similar picture, although expenditure on exceptional hardship payments was generally low many Housing Benefit managers considered their allocation to be 'about right' because it would allow them to cope with expected increases in demand in the future

Local authority benefit departments do not collect data on the number of restricted rent awards they make, but it is possible to get a sense from the quarterly returns that local authorities submit to DSS containing data on the number of Rent Officer restrictions in force in the relevant quarter

For the purposes of this research DSS supplied statistics on the number of Rent Officer restrictions in force in each local authority for the first three quarters of 1997/98 (Q1 to Q3). This allows us to get a sense of the order of magnitude of potential demand for exceptional hardship payments. For each authority, the number of Rent Officer restrictions in force in Q1, Q2, and Q3 was known. By taking the maximum value we have an indicator of potential demand in a particular quarter. This cannot be translated into the potential demand over the whole year because there are flows into and out of the register of restricted rents. Many claimants will appear in the statistics for all four quarters of the year Furthermore, claimants do not have to apply for a payment when their claim is determined. They may wish to apply at some point during the benefit period when they find they are in hardship. What we can sav however, is that the potential demand for exceptional hardship payments *could not be less* than the maximum figure for rent restrictions in the year.

The distribution of these maximum figures is presented in Table 3.5 below

| Maximum quarterly | Number of local | % | |
|-----------------------------|-----------------|----------|--|
| number of rent restrictions | authorities | | |
| 0-500 | 95 | 31 | |
| 501-1000 | 83 | 27 | |
| 1001-1500 | 39 | 13 | |
| 1501-2000 | 31 | 10 | |
| Over 2000 | 56 | 18 | |
| Total | 304 | <u> </u> | |

| Table 3 5 | Maximum quarterly rent restrictions as indicators |
|-------------|---|
| of potentia | l demand for exceptional hardship payments |

The purpose of this table is to make a simple point Table 3.5 suggests that the potential demand for exceptional hardship payments is very much higher than current application rates the data for which were presented earlier in Table 3.1 That table showed that only 20 per cent of local authorities received more than ten applications a month, whereas Table 3.5 suggests that most authorities *could* receive applications in the order of hundreds and sometimes thousands

These results are intriguing They suggest that the vast majority of claimants whose rents are restricted, and therefore do not receive the full amount of their rent in Housing Benefit, find some means of meeting the shortfall. A survey of local authority Housing Benefit managers cannot answer the question of how these claimants respond to having their rent met only in part rather than in full. Evidence from the development stage suggests that very few leave their accommodation as a result (since this would result in benefit awards ceasing and could therefore be identified from Housing Benefit records). However, from the face-to-face interviews with managers, front-line assessment staff, and fraud officers a large body of anecdotal evidence emerged. Below we list some of the reasons offered by local authority staff.

- shortfalls which are small can be met from claimants' income,
- some claimants negotiate a lower rent with their landlord,
- the rent cited on the Housing Benefit claim was fictitious, the landlord knowingly exaggerating the rent in an attempt to obtain a high level of Housing Benefit, but settling for what the local authority will pay.
- landlords accept whatever award of Housing Benefit is made in payment for rent, but treat the shortfall as arrears which is offset against the tenant's bond when they leave the property (This was viewed by local authority staff as a particularly cynical abuse of tenants by landlords)
- Discussion In the first two full years of the exceptional hardship payment scheme, the number of applications has clearly fallen well below the expectations of the DSS and local authorities The evidence is that applications are increasing but are still at a low level. From the data on Rent Officer restrictions there is seeningly a very large pool of potential applicants

There is some evidence from the survey that publicising the availability of exceptional hardship payments in a range of different ways, rather than relying solely on decision letters, could increase applications. Many authorities are using combinations of sometimes innovative methods to inform claimants about exceptional hardship payments. There may be scope therefore for a good practice guides containing examples of different wordings that could be used in decision letters and other forms of written information, and examples of more proactive methods of publicity. 4 WHY CLAIMANTS APPLY FOR EXCEPTIONAL HARDSHIP PAYMENTS AND HOW LOCAL AUTHORITIES MAKE DECISIONS

Introduction Local authorities have had, for many years prior to the exceptional hardship payments scheme, powers to pay additional Housing Benefit (i.e. above the maximum prescribed by regulations) in *exceptional arcumstances*. As explained in Chapter 1, from January 1996, when local reference rents were introduced, new powers were introduced enabling local authorities to pay an additional amount to bring Housing Benefit up to the level of a claimant's contractual rent (less ineligible charges) where in the assessment of the authority, the claimant or a family member would otherwise suffer *exceptional hardship*. In October 1996, the scheme was extended to cover people affected by the introduction of the 'single room rent' regulations.

As mentioned earlier, there is no definition of exceptional hardship in law but DSS guidance is contained in Circular HB/CTB A7/96. The circular makes it clear that local authorities have discretion to decide what constitutes exceptional hardship within 'its normal everyday meaning and usage'. However, some examples are given of factors which might be relevant. These include

- risk of eviction
- amount of shortfall between the claimant's rent and their Housing Benefit,
- claimant's other income (including disregarded income), assets and outgoings,
- possibility of claimant negotiating a lower rent,
- availability of alternative cheaper accommodation (including friends or relatives),
- the health of the claimant and/or family in relation to housing conditions
- effects on children of having to meet the shortfall,
- possibility of financial help from non-dependants in household,
- risk of (statutory) homelessness

The circular is clear that this list is not exhaustive Local authorities have discretion to consider any circumstances they see as appropriate and relevant

In this chapter we examine two related issues First, we look at the reasons put forward by claimants in their applications for exceptional hardship payments, and secondly we explore how local authorities have interpreted their discretionary powers and the mechanisms they use for making decisions

Why people apply for exceptional hardship payments From the development stage of the research, it was clear that, in their publicity materials, local authorities generally did not give examples of what might constitute *exceptional hardship* nor disclose how decisions will be made. It is left to the claimant to describe their circumstances in their own way. During this stage we collected numerous examples of the reasons put forward by claimants in their applications. For the survey, the reasons most frequently mentioned were listed and Housing Benefit managers asked to say whether, in their authority, the reason was common', not common or had not been used by any claimant'

Table 4.1 shows the percentage of Housing Benefit managers who said they had had examples of each reason in their authority

Table 4.1Claimants' reasons for requesting an exceptionalhardship payment, distribution among local authorities'

| Reason | Number of local authorities | % |
|---|-----------------------------|----|
| | which had heard of reason | |
| | in applications | |
| Claimant cannot afford rent | 286 | 99 |
| Claimant has medical condition or illness | 269 | 93 |
| Claimant has a disability | 254 | 87 |
| Claimant cannot afford to pay bil | Is 242 | 83 |
| Claimant is under 25 | 226 | 78 |
| Claimant needs room for childre | n 213 | 73 |
| to stay | | |
| Claimant is pregnant | 185 | 64 |
| Claimant cannot afford to pay | 179 | 62 |
| for food | | |
| Claimant is getting away from domestic violence | 36 | 47 |
| Claimant has language difficulties | 43 | 15 |

Housing Benefit managers were asked for examples of other reasons that they had come across in their local authorities. The following were mentioned

- claimant unable to find other suitable accommodation (mentioned by 13 local authorities)
- claimant needs to live close to family (12),
- claimant needs to live close to schools (12),
- claimant has other demands on their income (12)
- family member is ill or disabled (7)
- relationship breakdown involving children (7),
- eviction (6)
- ¹ Analysis excludes 15 local authorities who had not received in application in 1997/98

- claimant is victim of abuse (5),
- need to accommodate cater (2),
- claimant is drug-dependent (2),
- unfair treatment by landlord (2)
- claimant unable to find work (2)

Table 4.1 is based on data from all the authorities in the survey including those who had only had a small number of applications. To ask these authorities for examples of common reasons' is therefore inappropriate Table 4.2 is based only on those authorities who had 20 or more applications in 1997/98 and compares the proportion of Housing Benefit managers who had heard of the reason with the proportion who said the reason was 'common' in their authority.

| Reason P | roportion of local authorities | Proportion of authorities |
|-------------------------|--------------------------------|---------------------------|
| | which had heard of reason | where the reason |
| | in applications | was 'common' |
| Claimant cannot | 99 | 94 |
| afford rent | | |
| Claimant has medical | 97 | 74 |
| condition or illness | | |
| Claimant has a disabili | ту 96 | 58 |
| Claimant cannot afford | to 88 | 71 |
| pay bills | | |
| Claimant is under 25 | 84 | 46 |
| Claimant needs room | 84 | 36 |
| for children to stay | | |
| Claimant is pregnant | 79 | 18 |
| Claimant cannot affor | d 68 | 34 |
| to pay for food | | |
| Claimant is getting aw | ay 5 5 | 13 |
| from domestic violenc | e | |
| Claimant has language | 16 | 4 |
| difficulties | | |
| Number of authorities | s 160 | 160 |

Table 4.2 Analysis of 'common' reasons for applications inlocal authorities with 20 or more applications, 1997/98

Table 4.2 paints a slightly different picture to Table 4.1 Authorities which had higher levels of applications were clearly faced with a wide range of reasons on a regular basis. The high percentage of authorities saving that the mability to pay the rent was a common reason for applying for an exceptional hardship payment does not necessarily suggest that claimants misunderstand the scheme. It would certainly not be an adequate reason on its own for awarding an exceptional hardship payment but inability to pay the rent might have other consequences of more relevance. Indeed it was a finding from the development stage that applications for payments do not usually rely on one reason alone. Claimants generally

put forward a series of reasons which when put together presented a coherent narrative and plausible argument for exceptional hardship

The table is also interesting in that it highlights some of the sources of demand for private sector rented accommodation. Over one in ten of the authorities in this sub-sample said that women fleeing domestic violence was a common reason for applying for an exceptional hardship payment nearly one in five cited pregnancy, and over a third cited parents needing room for children to stay occasionally. There may be some idvantage, therefore, in providing advice to authorities about how the exceptional hardship payment scheme should be used in relation to circumstances such as these, which in the light of recent social trends might be expected to increase in number in future

Some of the reasons cited by claimants fall into distinctive groups. One group contains reasons related to the consequences that would follow from using other parts of the household budget to meet the shortfall in rent (including inability to pay for food or other bills) The second group imply the negative effects of having to give up the accommodation (including not being able to have children to stay, changing schools) Other reasons appear to be concerned more with exceptional (or difficult) circumstances rather than the consequences of having a shortfall in rent Here we could include illness and disability pregnancy, fleeing violence and relationship breakdown

As mentioned in the introduction to this chapter local authorities have wide discretion in deciding applications for exceptional hardship payments. hardship payments In the development stage however, there was a perception in some authorities that the regulations, as explained in DSS Circular A7/96, were actually somewhat restrictive. When the exceptional hardship payment scheme was first introduced in January 1996, the DSS guidance was the only assistance available to Housing Benefit departments. Since then many authorities have developed their own approaches to deciding claims, reflected in written policy documents, guidance to assessment staff application forms and publicity materials

> The variety of documentation that was identified in the development stage of the project suggested that there was no common way of describing them that would be recognised by all authorities. We were shown 'policy documents', minutes of meetings, training materials, and assessment proformas all of which were used to help decision makers. In the survey we devised a limited list which might cipture these differences and asked Housing Benefit managers to say if, in their authority, their approach to decision making was informed by one or more of the following

- a written policy document,
- a list of criteria to aid decision makers
- the DSS guidance circular

How local authorities make decisions on exceptional Table 4.3 presents the responses to this question – A combined category of policy/criteria is used in the table because of the difficulty of making a clear distinction between the two

| | Authorities use of aids to decision making (% of local authorities) | | | |
|----------------------|--|----------|---------|-----|
| | | | | |
| | Policy/ | DSS | Neither | |
| | criteria | guidance | | |
| English District and | 73 | 72 | 6 | 208 |
| Unitary authorities | | | | |
| English Metropolitan | 66 | 85 | 7 | 27 |
| authorities | | | | |
| London authorities | 71 | 59 | 18 | 17 |
| Welsh authorities | 64 | 57 | 29 | 14 |
| Scottish authorities | 32 | 73 | 14 | 22 |
| All authorities | 69 | 73 | 9 | 288 |

| Table 4.3 | Aids to decision making used by local authorities, |
|-------------|--|
| by local au | thority type |

The table shows that nearly three-quarters of the sample of authorities used the DSS guidance to assist in making exceptional hardship payment decisions. Almost as many had developed their own policy or set of criteria in addition or as an alternative. In contrast nearly one in ten authorities said that they used neither. Authorities were also asked whether they decided each case 'on its¹ments'. Without exception, all authorities answered that they did

Many authorities reported that they used both their own policy/criteria and the DSS guidance while others relied on one form of guidance or the other. It is possible therefore to construct a typology of authorities based on whether they used their own policy and whether they used the DSS guidance, as Figure 4.1 shows

| | | Use DSS guidance ² | | |
|------------------------------|-----|---------------------------------|-------------------------------|--|
| | | yes | no | |
| | | The 'comprehensive' approach | The 'independent' approach | |
| Have own policy criteria? | yes | 144 local authornies (50%) | 54 local authorities (19%) | |
| | no | The 'DSS-based' approach | The 'case-based' approach | |
| | | 65 local authornes (23%) | 25 local authornes (9%) | |

Figure 4.1 Decision making typology of local authorities (based on data from 288 local authorities)

Figure 4.1 shows that half the authorities in the sample used both their own policy/criteria and the DSS guidance. These authorities could be described as taking a comprehensive' approach to decision making. One in five authorities take a more independent line in using then own policy/criteria in preference to DSS guidance, but that does not imply that the guidance did not inform the authority's own approach. Nearly a quarter have been content to rely solely on the DSS guidance. Finally, almost one in ten authorities responded that they used neither their own policy nor DSS guidance. However, they did say that each case was considered on its merits indicating a distinctive 'case-based' approach

There is no intrinsic merit in any one of the approaches over the others, but the typology will be used in Chapter 5 to explore whether there is any evidence for a relationship between approaches to decision making and the outcomes of applications for exceptional hardship payments

Table 4.4 shows the distribution of local authority types among the cells of the typology matrix

| Type of authority | Local authorities' approach to | | | | | | | Base | | |
|----------------------|--------------------------------|------|-------|------|-------------|------|-------|------|-----|--|
| decision making | | | | | | | | | | |
| Com | prehensive | | DSS- | | Independent | | Case- | | | |
| | | | based | | | | based | | | |
| ii | No | (%) | No | (%) | No | (%) | No | (%) | | |
| English District and | 108 | (52) | 44 | (21) | 44 | (21) | 12 | (6) | 208 | |
| Unitary authorities | | | | | | | | | | |
| English Metropolitan | 16 | (59) | 7 | (26) | 2 | (7) | 2 | (7) | 27 | |
| authorities | | | | | | | | | | |
| London authorities | 8 | (47) | 2 | (12) | 4 | (24) | 3 | (18) | 17 | |
| Welsh authorities | 7 | (50) | I | (7) | 2 | (14) | 4 | (29) | 14 | |
| Scottish authorities | 4 | (18) | 12 | (55) | 3 | (14) | 3 | (14) | 22 | |
| All authorities | | (50) | | (23) | | (19) | | (9) | 289 | |

| Table 4.4 | Approaches to decision making by local authority |
|-----------|--|
| type | |

Some caution must be exercised in interpreting this table because of the relatively small base numbers for local authority types other than the English districts. Nevertheless, there are a number of observations that can be made

- reliance on DSS guidance appears greater in Scotland than in other authorities,
- · London and Welsh authorities rely on DSS guidance the least,
- · the case-based approach is used most in Wales,
- the independent approach is used most in London,
- more Metropolitan authorities adopt the comprehensive approach than other types of authority

Relevance of pre-tenancy determinations in decision making The relevance of pre-tenancy determinations for exceptional hardship payments is that there might be an expectation that claimants who know (from a pre-tenancy determination) that they are unlikely to receive Housing Benefit which will meet their rent in full would either accept the tenancy and be prepared to meet the shortfall or look for alternative accommodation. In the survey, Housing Benefit managers were asked if the existence of a determination played a part in deciding an application for an exceptional hardship payment. Table 4.5 presents their responses

Table 4.5Relevance of pre-tenancy determinations fordeciding applications for exceptional hardship payments, bylocal authority type

| | Is PTD r | elevant? | Base (number | |
|--|--------------------|----------|-----------------|--|
| | (% of authorities) | | of authorities) | |
| | Yes | No | | |
| English District and Unitary authorities | 92 | 8 | 208 | |
| English Metropolitan authorities | 93 | 7 | 27 | |
| London authorities | 88 | 12 | 17 | |
| Welsh authorities | 100 | 0 | 14 | |
| Scottish authorities | 82 | 18 | 22 | |
| All authorities | 91 | 9 | 288 | |

As Table 4.5 shows the large majority of authorities take pre-tenancy determinations into account when deciding applications for exceptional hardship payments. This finding also reflects the findings from the development stage of the project. Many of the managers interviewed during that stage emphasised that a pre-tenancy determination was relevant but only one of a range of relevant factors. The existence of a determination would not be a sufficient reason to reject an application

determination would not be a sufficient reason to reject an application At the mid point of the telephone survey, we identified local authorities where a written policy or list of criteria was in use. These 161 local authorities were sent a request for copies of their documents, of which

The documents that were returned by local authorities varied enormously in their style and content. Many were not policy documents in the sense of a clear statement of how the authority viewed the exceptional hardship payment scheme. Many documents were application forms that claimants were required to complete. However, it is probably fair to say that the approach of the authority will be reflected in the forms in the scope and detail of the questions asked.

There are a number of observations that can be made from studying the policy documents, lists of criteria, application forms and other material

• many application forms were long and complicated, some stretching to many pages containing perhaps 20 or more questions

Content analysis of local authorities policies/criteria

88 responded

<u>29</u>

- often the questions on forms were clearly based on the examples of possible relevant factors found in Circular A7/96 (such as have you tried to renegotiate the rent with your landloid? or what other properties have you looked at²)
- many authorities required information about claimants expenditure This was often in the form of a list of expenditure items to complete These commonly covered essential items such as food, fuel, water clothing, and expenditure on repaying debts and loans. Some inthorities asked for other details such as expenditure on cars, telephones or televisions, and leisure spending like eigarettes and alcohol. The longest expenditure form contained 68 items for the claimant to complete (including presents, children's pocket money) newspapers and prescriptions) and an additional form requiring details of creditors,
- a number of authorities had developed scoring systems for helping them assess the strength of a claimant's application,
- some authorities pointed out that their assessment procedures included visiting the claimant in their own home

Local authorities had a ringe of different approaches to what constituted exceptional hardship. In one authority, the working definition in their guidance notes to staff included the requirement that a claimant *must* be facing eviction from their accommodation. Another authority took a dictionary definition as its guide (and mentioned this in correspondence to claimants). It was 'exceptional means' something unusual or uncommon' while 'hardship' means 'severe suffering, extreme privation From a scrutiny of the documents supplied by local authorities, the following list of factors has been identified that are additional to those included in the DSS guidance Circular HB/CTB A7/96 (set out at the beginning of this chapter).

- a recent death in the household,
- · special dietary heiting or laundry requirements,
- whether claimant had applied for local authority or Housing Association accommodation,
- reasons for choosing current accommodation,
- · how the claimant found out about the current accommodation,
- reasons for requiring 'over-large' rooms,
- reasons for claimant's financial position.
- possibility of a third party negotiating with the landlord,
- possibility of claimant taking on a sub-tenant
- action claimant will take if exceptional hardship payment is not awarded
- likelihood of changes in circumstances in the near future.
- possible income from social security benefits not claimed,
- possibility of assistance from Social Services Department.

It is clear from the documents collected in the course of the project that local authorities have from the common starting point of Circular A7/ 96 developed their approach to decision making in very different ways At this stage at is not possible to draw conclusions from this diversity, but one specific concern can be taised. It is possible that many potential applicants will be put off subnutting an award by the scale and detail of the information they are being asked to supply particularly about their expenditure. Also questions about spending on personal items, such as prescriptions pocket money, presents and cigarettes, could be construed as unnecessarily intrusive.

Discussion As Chapter 3 demonstrated, the large number of Rent Officer restrictions on private sector rents has created a large pool of potential applicants for exceptional hardship payments whose circumstances will vary widely. It is not surprising therefore to find that local authorities encounter such a wide range of reasons in the applications they receive

> Many of the reasons cited by applicants appear to relate to the circumstances which gave rise to their claim for Housing Benefit, in other words, why they are in rented accommodation with a restricted rent. Other reasons appear to describe generally difficult circumstances, such as illness or disability, which can result in financial pressures on people with low incomes. It seems therefore that the range of factors which the DSS foresaw as being possibly relevant is only partially reflected in the reasons put forward by claimants. For example, few authorities had come across claims that eviction or homelessness would be the result of Housing Benefit not meeting the rent in full. In general, applications appear to arise primarily from the fact that claimants have to find some part of the rent from their income.

> The results from the survey suggest that most authorities (around seven in ten) have chosen to devise some form of guidelines other than Circular A7/96 to assist their assessment staff. Around half have adopted a 'comprehensive' approach by using both. The scope and content of these 'policy documents' appear, from the examples sent by 88 local authorities, to vary enormously between authorities. Particularly striking was the number of authorities who require applicants to complete detailed expenditure forms as part of their application. It is possible that these could discourage applications. The most sophisticated approach to decision making was found in five local authorities which had devised (sometimes elaborate) scoring systems for assessing applications. (Three of these authorities had success rates for claimants making applications of over 60 per cent.)

5 OUTCOMES OF APPLICATIONS FOR EXCEPTIONAL HARDSHIP PAYMENTS

| Introduction | In this chapter we examine the outcomes of applications for exceptional |
|--------------|--|
| | hardship payments. Successful applications are analysed by the type of |
| | claimant making the application, and by the value of the awards made |
| | Unsuccessful applicants are analysed by claimant group to explore the |
| | possibility that claimant groups fare differently when they apply Finally, |
| | the number and outcomes of appeals against adverse decisions are analysed |

The data used in this chapter derive from the survey – Housing Benefit managers were asked for the following statistics for 1997/98

- · total number of exceptional hardship phyments awarded,
- number of exceptional hardship payments awarded by claimant type
- number of unsuccessful applications
- · number of unsuccessful applications by claimant type,
- · lowest weekly amount of exceptional hirdship payment awarded
- highest weekly amount of exceptional hardship payment hwarded
- · average weekly amount of exceptional hardship payment awarded,
- · number of appeals number of successful appeals

Collecting data for the survey was not straightforward for local authorities Almost every local authority kept records (in varving amounts of detail) of the number and the amounts of awards and payment periods. Most kept these on computer as a means of tracking payments and monitoring ongoing expenditure. Information on failed applications was not routinely held and in many cases had to be extracted manually from casepapers. Some authorities did not inswer the questions on unsuccessful applications because they did not have the resources to do so. For these authorities it has not been possible to calculate numbers of applications of success rates

Successful applications and O. success rates for Successful applications O

Of the 305 local authorities in the survey, 288 authorities supplied figures for the number of exceptional hardship payments awarded in 1997/98 Of these, 248 were also able to supply figures for the number of unsuccessful applications. Aggregating the responses from these authorities shows that 9-483 exceptional hardship payments were made from 22/034 applications, an overall success rate of 43 per cent. The mean number of payments in these authorities was 38 within a range of between one and 664

The distribution of the number of awards between different authority types is shown in Table 5-1

| | Excep | tional hard | lship paym | ients award | ed | Total |
|----------------------|-------|-------------|------------|-------------|------|-------|
| | | (num | ber of aut | horities) | | |
| | Zero | 1-10 | 11-40 | 41-100 | 101+ | |
| English District and | 15 | 89 | 71 | 27 | 14 | 216 |
| Unitary authorities | | | | | | |
| English Metropolitan | 0 | 6 | 8 | 7 | 6 | 27 |
| authorities | | | | | | |
| London authorities | I | 2 | 5 | 6 | 4 | 18 |
| Welsh authorities | 0 | 6 | 5 | 3 | 0 | 14 |
| Scottish authorities | 2 | 12 | 6 | 2 | I | 23 |
| All authorities | 18 | 115 | 95 | 45 | 35 | 288 |
| | (6%) | (39%) | (32%) | (15%) | (8%) | |

Table 5 1Number of exceptional hardship paymentsawarded, by local authority type

The raw data have been used in this table principally because the cell sizes are too small for percentage figures to be meaningful. Furthermore, because three-quarters of the authorities in the sample are English Districts, the percentage figures in the bottom row for all authorities are largely only a reflection of the distribution of the Districts. Nevertheless the table does present a picture of widely differing experiences between authorities. As might be expected, the general pattern for the larger London and Metropolitan authorities is different from the general pattern for the smaller authorities in Wales and Scotland. That 45 per cent of authorities had made ten or fewer exceptional hardship payments in 1997/98 (including 18 authorities which had made no payments) suggests that the scheme is probably not working as expected.

There are several possible contributory explanations for the apparently low number of exceptional hardship payments in some authorities including

- more claims are being rejected than allowed
- the criteria used for decision making tend to exclude applicants rather than include them,
- local authorities may not making many awards in order to reduce their own costs and to keep the central government contribution to the budget
- claimants are mostly able to meet the shortfall in their Housing Benefit from their own resources or negotiate a lower rent with their landlord

The data from the survey allows us to explore the first two of these possibilities. We cannot make any systematic assessment of the third and fourth points, however

Success rates Success rates were calculated for the 157 authorities with 20 or more applications in 1997/98 Table 5.2 shows the results of this analysis. Except

for English District authorities the numbers of local authorities in each category were too small to calculate percentages. The raw data are therefore shown for all authorities.

| | Success rates for exceptional hardship payment applications (number of authorities) | | | | |
|----------------------|---|--------|--------|---------|-----|
| | | | | | |
| | | | | | |
| | 0-25% | 26-50% | 51-75% | 76-100% | |
| English District and | - 19 | 29 | 39 | 23 | 110 |
| Unitary authorities | | | | | |
| English Metropolitan | 9 | 5 | 3 | I | 18 |
| authorities | | | | | |
| London authorities | 5 | 2 | 3 | 3 | 13 |
| Welsh authorities | 2 | 3 | 3 | 0 | 8 |
| Scottish authorities | I | 2 | 2 | 3 | 8 |
| All authorities | 36 | 41 | 50 | 30 | 157 |
| | (23%) | (26%) | (32%) | (19%) | |

| Table 5.2 Success rate of applications, by local authority t |
|--|
|--|

Table 5.2 suggests that there is a wide ringe of fairly evenly distributed success rates across the country. There appears to be no intrinsic reason why some authorities granted most of their applications for exceptional hardship payments (or *all* applications in four local authorities), while others rejected most. One possible contributory explanation might be that local authorities have very different standards by which they assess claims, some taking a presumably tight approach to the definitions of the words 'exceptional' and hardship , and others taking a far wider and more inclusive interpretation.

Table 5.3 compares the overall success rates of applicants with the approach taken by authorities to decision making as identified in the previous chapter. The 'success rate' cited in the table is calculated from the total number of applications and the total number of successes across all the authorities in each of the categories of approaches to decision making

| Type of decision making approach | Overall success rate | Base number of |
|---|----------------------|----------------|
| | of applications | local |
| | | authorities |
| Comprehensive (ie based on own policy | 43 | 131 |
| and DSS guidance) | | |
| DSS-based (using DSS guidance only) | 51 | 47 |
| Independent (using own policy only) | 50 | 48 |
| Case-based (no policy, DSS guidance not used) | 30 | 21 |
| All authorities | 43 | 247 |

Table 5.3 Overall success rates compared with decision making approach of local authorities

Although the number of local authorities using the case-based' approach is small the table suggests that this approach where each case is considered on its merits without reference to any form of guidance may lead to fewer awards than other approaches. The table also raises the question of whether decision makers with more than one source of guidance (the 'comprehensive' category) in practice have to consider more factors in making a decision than those who rely on either the DSS guidance of the authority's own policy. The effect of devising an in-house policy may have been to decrease the likelihood that a clumant will satisfy the criteria for exceptional hardship'

One of the questions that the research set out to explore was whether particular types of claimant tended to be more or less successful in applying for an exceptional hardship payment than others. In the survey, therefore Housing Benefit managers were asked to break down successful and unsuccessful applicants into the main claimant groups of lone parents, pensioners and disabled people. Although the administrative classification of claimants tends to treat these groups as mutually exclusive, it is possible for a clamant to fall into more than one. Hence, the analysis presented here is used only to indicate broad brush differences between them rather than statistically significant findings - Housing Benefit managers were also asked how many applicants were single people under 25 pregnant women, people with mental health problems or absent parents (again recognising that these are overlapping categories). This sort of information is not collected by local authorities in any systematic way although the relevant information is usually contained within the case record. Some authorities were not able to supply the data required others supplied estimates only. The data are therefore limited, but it is possible to get a sense of whether different claimant groups fare better than others in applying for exceptional hardship payments. Using data from only those authorities which could supply breakdowns of both successful and unsuccessful applicants we can calculate an aggregate success rate for each claimant group shown in Table 5.4

Analysis of exceptional hardship payment recipients

| | Number of | Number of | % | Base |
|-----------------|----------------------|--------------|---------|--------------|
| | exceptional hardship | unsuccessful | success | (number of |
| | payments made | applications | rate | authorities) |
| Lone parents | 1585 | 1625 | 49 | 221 |
| Disabled people | 807 | 377 | 68 | 220 |
| Pensioners | 295 | 306 | 49 | 218 |

| Table 5.4 Success rates of different claimant groups | Table 5.4 | Success | rates of | different | claimant gr | oups |
|--|-----------|---------|----------|-----------|-------------|------|
|--|-----------|---------|----------|-----------|-------------|------|

The table indicates that disabled people are more likely to succeed with an application for an exceptional hardship payment than either lone parents or pensioners. Nevertheless lone parents were awarded nearly twice as many payments as disabled people and five times as many as pensioners. It is not possible from the survey data to explore the reasons for these differences. Clearly the number of awards will be related to the number of applications from each of the claimant groups which in turn will be related to how many of each type of claimant is in the Housing Benefit population. Also, from what is known about take-up rates for meanstested benefits generally. It is likely that numbers of applications from pensioners, will be considerably lower than for lone parents, for example. The table does indicate, though, is that the principal beneficiaries from the exceptional hardship payment scheme are lone parents.

It was also possible from the survey data to explore the relative success rates for claimants with particular characteristics (rather than as members of specific claimant groups) Table 5.5 presents the results of this analysis

| | Number of | Number of | % | Base |
|---------------------|----------------------|--------------|---------|--------------|
| | exceptional hardship | unsuccessful | success | (number of |
| | payments made | applications | rate | authorities) |
| Single claimant | 821 | 1010 | 45 | 223 |
| under 25 | | | | |
| Claimant with menta | 1 339 | 101 | 77 | 213 |
| health problems | | | | |
| Pregnant women | 173 | 68 | 72 | 215 |
| Absent parents | 55 | 48 | 53 | 214 |

Table 5.5Success rates of claimants with specificcharacteristics

It is interesting to compare this table with Table 5.4. It appears that in 1997/98 single people under 25, whose Housing Benefit will have been subject to the restrictions imposed by the 'single room rate', applied for exceptional hardship payments in comparable numbers to disabled people and far more than pensioners. Their success rate (45 per cent) was slightly lower than for lone parents and pensioners (both 49 per cent). Applications

by pensioners were outstripped by applications from people with mental health problems, who according to Table 5-5 secured the highest success rate of all types of applicant. Pregnant women were also relatively successful in getting exceptional hardship payments, although there were few of these. Absent parents represented a small fraction of applicants but were successful in over half their applications.

Value of exceptional hardship payments

We know from the development stage of the project that the value of exceptional hardship payments varies enormously depending on the shortfall between a claimant's rent and their Housing Benefit. Some claimants are awarded only modest amounts, sometimes as a contribution to the shortfall rather than the full amount, while the circumstances of other claimants justify very large payments. In the survey, we asked Housing Benefit managers the values of the lowest and highest weekly payments made in their authorities. This allows us to establish the *range* within which payments were made in 1997/98. We also asked for an estimate of the average weekly payment. Table 5.6 presents the results from the analysis of lowest and highest weekly payments.

| | Percentage of local authorities |
|-------------------------------|---------------------------------|
| Value of lowest weekly award | |
| £1 5 | 73 |
| £6 10 | 17 |
| £11 20 | 6 |
| Over £20 | 3 |
| Base number of authorities | 275 |
| Value of highest weekly award | |
| LI-25 | 36 |
| £26 50 | 44 |
| £5 100 | 17 |
| Over £100 | 4 |
| Base number of authorities | 275 |

Most authorities made some low weekly payments as Table 5.6 shows. This finding raises an interesting question. Can payments of a few pounds be a plausible response to a claimant in apparently exceptional hardship? In other words, what exceptional hardship could result from a claimant having to find only one or two pounds from their other resources? Several valid responses can be made to such questions. First, during the development stage, we were given examples of claimants whose budgets, on a very low income, were so tight that having to find a shortfall of a few pounds would have had serious consequences for the claimant's ability to pay for example, for food or essential services. Secondly, it must be

4

remembered that local authorities do not have to meet the shortfall in full but have the discretion to make any level of payment up to the full amount. The small amounts that are sometimes paid could therefore be in relation to cases where the local authority has decided to make a contribution to the shortfall. Several managers explained that when there is a large shortfall it is often the case that the claimant is able (and willing) to meet some of the extra cost but that exceptional hardship would result if they were forced to pay the full amount. In such circumstances an appropriate response is to make an exceptional hardship payment at an amount that will prevent exceptional hardship but no more

Table 5.6 also shows that over one in five authorities made some relatively high weekly payments, in excess of $\pounds 50$, and a small number made payments of over $\pounds 100$ per week. However, these are likely to be isolated cases with particularly unusual sets of circumstances, since, as Table 5.7 shows, almost all authorities reported average weekly payments of $\pounds 30$ or less.

| Table 5.7 | Average weekly | exceptional | hardship | payments |
|------------|------------------|-------------|----------|----------|
| made by lo | ocal authorities | | | |

| Value of overage weekly award | Percentage of local authorities |
|-------------------------------|---------------------------------|
| £1-10 | 33 |
| £11-20 | 50 |
| £21 30 | 15 |
| Over £30 | 2 |
| Base number of authorities | 275 |
| | |

From the development stage of the project it was clear that the overwhelming majority of applications for exceptional hardship payments came at the start of a claim, when a claunant moved into a new property rather than as a result of a change in circumstances once a Housing Benefit awind was in payment. Exceptional hardship payment cases then tended to fall into one of two types.

- Cases where the nature of the exceptional hardship was considered temporary, which would be removed when the claimant found more suitable accommodation
- Cases where the nature of the exceptional hardship was considered permanent and which could not be alleviated by a move to other accommodation

The first type of case typically included claimants who had had to move into new accommodation as a matter of some urgency for example as a result of a relationship breakdown involving children or because of domestic violence. In such cases, exceptional hardship payments were usually awarded for a period of time long enough to enable the claimant to find alternative accommodation without adding to the pressure that they might already be under. The second type of case typically included claimants who had some form of special needs which were ongoing rather than temporary and which required possibly large or high quality accommodation. Another example is the claimant who takes accommodation in a particular (usually expensive) area for some family or domestic reason, such as to provide care for a relative of to be near a particular school.

In the first type of case an exceptional hirdship payment might be awarded for something like three to six months. In the second type of case a payment might be continuous over several benefit periods. In either case the total value of an exceptional hardship payment to an individual claimant can be substantial. From the survey data, it was possible to calculate a rough indicator of the magnitude of the amounts paid to individual claimants, using the total expenditure on exceptional hardship payments and the number of successful claims. Table 5.8 presents the results of this estimation

| Estimated average value | Number of local | Percentage | |
|-------------------------------|-----------------|------------|--|
| exceptional hardship payments | authorities | | |
| to individual claimants | | | |
| £+ 250 | 78 | 28 | |
| £251-500 | 114 | 42 | |
| £501-750 | 43 | 16 | |
| £751-1000 | 18 | 7 | |
| over £1000 | 21 | 8 | |
| Base number of authorities | 274 | | |

Table 5.8 Estimate of average value of exceptional hardshippayments to successful applicants

Tables 5.7 and 5.8 present a consistent picture of the majority of exceptional hardship payments being at the lower end of the range (up to $\pounds 20$) which if paid for a period up to six months would accrue to the claimant a total amount up to $\pounds 500$. There are also some authorities where claimants are receiving much larger amounts. In particular Table 5.8 includes 21 authorities where claimants are receiving on average, over $\pounds 1000$ from exceptional hardship payments.

Appeals As mentioned earlier, the survey data can be used to produce a success rate for exceptional hardship payment applications of 43 per cent. Hence over half of all applications end in failure – Like all Housing Benefit decisions a retusal to award an exceptional hardship payment can be appealed by the claimant – In the survey data on appeals were provided by 228 local authorities. In these authorities, 3-272 appeals were lodged, an appeal rate of around 26 per cent. One authority was an outher reporting 750 appeals. If we remove this authority from the calculation we arrive at a revised appeal rate of 21 per cent. Of all 3-272 appeals, 822 were successful a success rate of 25 per cent / However removing the outlier authority produces an amended success rate of 30 per cent (a figure comparable to the success rate of means-tested social security benefits)

Most authorities had dealt with verv few appeals Sixty-two (27 per cent) reported that they had not received any appeals, a further 96 (42 per cent) had received between one and five. Only one *in* five authorities had had more than ten appeals. Table 5.9 shows the distribution of appeals among local authority types. Except for English District authorities, the numbers of local authorities in each category were too small to calculate percentages. The raw data are shown instead.

| | | Numb | er of app | eals | | | Base |
|----------------------|-------|-------|-----------|-----------|-------|------|------|
| | | (numb | er of aut | horities) | | | |
| | Zero | 1-5 | 6-10 | 11-20 | 21-50 | 51+ | |
| English District and | 52 | 77 | 20 | 8 | 9 | 4 | 170 |
| Unitary authorities | | | | | | | |
| English Metropolitan | 0 | 5 | L | 2 | 5 | 5 | ١ß |
| authorities | | | | | | | |
| London authorities | 0 | 4 | (| i i | 2 | 4 | 12 |
| Welsh authorities | 4 | 3 | 2 | 3 | 0 | I | 13 |
| Scottish authorities | 6 | 7 | 2 | 0 | 0 | 0 | 15 |
| All authorities | 62 | 96 | 26 | 14 | 16 | 14 | 228 |
| | (27%) | (42%) | (11%) | (6%) | (7%) | (6%) | |

Table 5.9 Number of appeals, by local authority type

Although the cell sizes in the table are generally small, there is an indication that the larger authorities in entitle London and Metropolitan authorities received more appeals than the smaller authorities in England, Scotland and Wales. There wis also a small number of authorities where appealing against refusals to awrid exceptional hardship payments was in comparison with most other authorities, a relatively common experience, in the order of at least one a week

Appeal rates between individual local authorities varied widely Table 5.10 shows the distribution of appeal rates for those authorities with 20 or more rejected applications for an exceptional hardship payment

| Appeal rates | Percentage of authorities |
|----------------------------|---------------------------|
| <10% | 40 |
| 11 20% | 23 |
| 21 30% | 14 |
| 31-50% | 18 |
| >50% | 6 |
| Base number of authorities | 106 |

Table 5 10 Appeal rates within individual authorities with more than 20 rejected applications (base = 106 authorities)

The table shows the wide range of appeal rates among individual authorities. The data from the survey do not allow us to explore the reasons for the variation but relevant contributors factors might include the personal motivation of claimants, the amount of exceptional hardship payment involved, the availability of alternative accommodation for claimants, and welfare rights services (their number, availability and effectiveness). Nevertheless, it is interesting that of the 25 authorities in the higher two bands (i.e. over 30 per cent) three are London Boroughs and six are coastal authorities (in England, Scotland and Wales).

Appeals against Housing Benefit decisions are decided in the first instance by an internal administrative review. Unsuccessful claimants have a further right of appeals to a Housing Benefit review Board comprising elected local authority councillors. In the authorities visited in the development stage, very few had held a single Review Board. This finding is in line with findings from previous research (Samsbury and Eardlev, 1991) that few unsuccessful claimants appeal beyond the internal review.

Discussion Chapter 3 demonstrated the wide variation between authorities in the number of applications for exceptional hardship phyments in 1997/98. This picture of wide variation is repeated when we look at the number of awards that are made and the success rates of applicants. Why some authorities should apparently have success rates in single figures while others inde payments to every claimant who applied is puzzling. What is clear, however, is that which local authority you live in appears to have a strong bearing on the likelihood of your application for an exceptional hardship payment succeeding. The comparison of success rates with authorities approaches to decision making also suggests that how decisions are made (i.e. what forms of guidance are used) could have a bearing on whether applications succeed or not

The suggestion that the DSS guidance is restrictive and imposes tough criteria on applicants was made by some of the Housing Benefit staff interviewed in the development stage of the project. The results from the survey on this point are inconclusive. Success rates for claimants were highest in authorities relying on the DSS guidance and comparable to

j

those in authorities which used their own policy (51 and 50 per cent respectively). We can interpret this finding in different ways. It may be that local authorities, own policies are equally as restrictive as the DSS guidance and therefore excluding some claimants in the same way. It may ilso be possible that the guidance is not as restrictive as suggested by the managers interviewed in the development stage.

From the survey data we can infer that people whose circumstances are related in some way to their health (disabled people, those with mental health problems, or pregnant women) are the most successful types of claimant. However, the most numerous beneficiaries of exceptional hardship payments were lone parents. Pensioners appeared to fare badly in the sense that applications were seemingly low compared with the large number of pensioners in the population.

Wide variations between local authorities were also apparent in the range of exceptional hardship payments made and the average amounts paid to individual claimants. In some authorities apparently large weekly payments and large aggregate amounts were the norm rather than the exception

Relatively few unsuccessful applicants pursued their cases to appeal (or internal review in the first instance). The success rate of about one in four is in line with success rates for other means-tested benefits.

6 ADMINISTERING THE SCHEME

| Introduction | Implementation of the exceptional hardship payment scheme has been | | | | |
|--------------|--|--|--|--|--|
| | undertaken in various ways in different authorities. Arrangements for | | | | |
| | publicising, deciding applications, dealing with appeals, recording and | | | | |
| | monitoring have all been required. From the development stage of the | | | | |
| | project, it was apparent that most authorities used their computer systems | | | | |
| | to some degree to keep records and manage expenditure | | | | |

Views about administration In the survey Housing Benefit managers were asked the extent to which administration of the scheme was easy or caused difficulties. Table 6.1 presents the responses by local authority type

| Type of authority | y Experience of administration | | | | | Base | | | |
|----------------------|--------------------------------|------|----|------|--------|--------|------|------|-----|
| | by local authorities | | | | | | | | |
| | Ve | ery | Ma | inly | So | me | Ser | lous | |
| | ez | isy | ea | sy | diffic | ulties | prol | blem | |
| | No | (%) | No | (%) | No | (%) | No | (%) | |
| English District and | 15 | (7) | 99 | (47) | 92 | (44) | 4 | (2) | 210 |
| Unitary authorities | | | | | | | | | |
| English Metropolitan | 0 | (0) | 8 | (30) | 18 | (67) | ł | (4) | 27 |
| authorities | | | | | | | | | |
| London authorities | I | (6) | 7 | (41) | 8 | (47) | Ţ | (6) | 17 |
| Welsh authorities | ł | (7) | 6 | (43) | 6 | (43) | T. | (7) | 4 |
| Scottish authorities | 3 | (14) | 8 | (47) | 8 | (41) | I. | (5) | 22 |
| All authorities | | (7) | | (44) | · | (46) | | (3) | 290 |

Table 6.1 Experience of administering the exceptionalhardship payment scheme, by local authority type

The table shows that just under a half of the authorities in the survey have experienced some difficulties in the administration of exceptional hardship payments and that a small percentage (eight authorities) reported that they had had 'serious problems'. In the survey, Housing Benefit managers were invited to explain the nature of the difficulties that they had encountered. The most common among the 142 respondents were as follows

- lack of guidance on what constitutes 'exceptional hardship' (mentioned by 51 local authorities),
- computer problems, including setting up systems and with software (43)
- ensuring staff were sufficiently trained and equipped (22),
- · length of time needed to investigate applications (20),
- keeping track of cases and expenditure (18),

- amount of work involved gathering information (17)
- informing claimants of new provisions (15)
- concerns about exceeding the budget (8),
- using manual systems (5)

When we looked at the responses from the eight local authorities which reported serious problems we found no consistent pattern in the types of difficulties described. Interestingly only one of the eight mentioned that they had had computer problems

Table 6.2 compares local authorities' experiences of administering exceptional hardship payments with the rate of applications they received

| Number of applic | ations Expe | erience of adr | ninistration | | Base |
|------------------|-------------|----------------|--------------|---------|------|
| per month | | (% of autho | | | |
| | Very | Mainly | Some | Serious | |
| | easy | easy | difficulties | problem | |
| <1 | 14 | 63 | 23 | 0 | 64 |
| >1-5 | 4 | 44 | 50 | 2 | 113 |
| >5-10 | 11 | 37 | 48 | 4 | 27 |
| >10 | 6 | 28 | 57 | 9 | 54 |
| All authorities | 7 | 45 | 45 | 3 | 258 |

Table 6.2. Experience of administration, by rate of applications

The inference that can be drawn from this table is that the local authorities with the higher rates of applications tended to report more problems with implementation than those with lower application rates. This suggests that it is possibly the process of routinely having to deal with often difficult decisions that local authorities have experienced as problematic rather than say, the task of introducing appropriate computer systems. The types of problem cited by local authorities (listed above) tend to add support to this interpretation since most are concerned with decision making processes. The 43 authorities who reported computer problems represent only 15 per cent of the 290 authorities which have implemented the scheme (in the sense of having any applications to deal with)

1

Discussion The question in the survey interview did not ask specifically about setting up the scheme but referred more widely to local authorities' experiences of administering exceptional hardship payments. As mentioned above many of the problems described were about day-to-day administrative issues (such as resource management, training, and monitoring) that are essentially within the power of local authorities themselves to alleviate rather than being issues of policy. The data from the survey suggest that few authorities experienced serious difficulties in setting up the exceptional hardship payment scheme. The lack of problems may in part be the result of the generally low take-up of payments in the first year and since One of the enduring problems (mentioned by over 50 Housing Benefit managers) concerns the interpretation of what constitutes 'exceptional hardship'. This difficulty was also mentioned in some of the local authorities in the development stage of the project. Some authorities would clearly welcome more detailed advice and guidance. Equally others are comfortable with the broad framework of exceptional hardship payments which allows them the flexibility to devise local policies which fit with the general policy direction of benefits in their authority.

7 EXCEPTIONAL HARDSHIP PAYMENTS - SUMMARY AND DISCUSSION

One of the main objectives of this research into local authorities discretionary powers was to investigate how the exceptional hardship payments scheme is operated by authorities and, in particular why expenditure has been lower than expected while at the same time varying so significantly between them

The research was designed to explore a number of specific questions which were set out in Chapter 1. In this concluding chapter, we draw on the findings of the study to address each of these. Finally, we offer some further comments on possible policy development in this area

The research questions Why do local authorities spend very different amounts on exceptional hardship payments?

How many applications are made, and how many are successful?

What are the patterns of applications and awards for exceptional hardship payments between local authornies?

These questions are linked but there is clearly no simple answer to any of them. In some ways it would be surprising if local authorities, given the diversity of their geography demography political control and other factors, displayed consistency in the way they administered exceptional hardship payments. The current research can contribute to an understanding of why expenditure patterns are so different between authorities but there is a range of other factors which affect spending which the research could not address. A provisional list of the principal factors would include the following some of which are independent and some of which are related

1 Demand

Expenditure on exceptional hardship payments is, in the first instance, demand-led. No applications would mean no expenditure regardless of any other factors. Demand will be influenced by factors which contribute to levels of take-up, and non-take-up of benefits such as knowledge, value, perceived need, hassle and the monetary and social costs to the claimant. Claimants' knowledge about exceptional hardship payments will in turn be influenced partly by the amount of information provided by the local authority in its letters and other publicity material. The level and effectiveness of local advice and advocacy agencies will also affect demand. The responses of landlords to tenants whose Housing Benefit falls short of the asking rent will also be relevant.

2 Need

Exceptional hardship payments are pavable only to people whose Housing Benefit does not meet their actual rent. The number and size of the shortfalls depend on the implementation of local reference rents and single room rents by the Rent Officer

3 Local authority policy

The exceptional hardship payment scheme is discretionary. I ocal authorities have the DSS guidelines but are also able to devise their own policies for how the funds are spent. For example, there may be a policy on whether shortfalls are met in full or whether a contribution only is made instead. The length of time for which payments are made may also be a matter of policy.

4 Housing Benefit management

Housing Benefit managers will have responsibility for controlling expenditure on exceptional hardship payments. They are likely to monitor or control the budget. How they respond to this responsibility will influence front-line decision making. For example, they may limit expenditure early in the year for fear of overspending later.

5 Front line decision making

Decisions are made by a range of different grades of staff How they interpret guidelines, local policy etc and the extent to which they are allowed to use their own discretion will influence outcomes

It can be seen from this list there is no simple relationship between, for example, the size of a local authority's Housing Benefit caseload or the amounts paid out to tenants in private sector accommodation, and the amounts paid by a local authority in exceptional hardship payments. Nevertheless, despite the complexity of the picture presented above, the research can offer some suggestions as to why the variations in exceptional hardship payment expenditure are so wide. Expenditure is principally dictated by the number of awards made and the amounts of those awards. Local authorities have been shown to differ widely in the number of awards they make (from zero to many hundreds) and in the amounts they pay (some appatently making modest payments and others occasionally making very large awards).

The number of awards is linked in part to the number of applications made by Housing Benefit claimants, and again local authorities varied widely in the number they received. The question is therefore raised of why application rates vary so much (as discussed in Chapter 3). The research suggests two contributory reasons. First, local authorities publicise the availability of exceptional hardship payments in different ways. Some take the simple but minimal, approach of informing claimants in letters notifying claimants of the outcome of their claim. Others use different methods in addition, such as providing information to local advice agencies, which the research indicates is associated with higher levels of applications.

The number of awards is also linked to the way authorities decide claims Analysis of the success rates of applicants shows once again that local authorities differ widely in the ratio of applications they allow to those they reject. It is possible that some authorities attract mostly claims which are likely to be successful. This scenario nught arise for example, if the publicity for exceptional hardship payments effectively discouraged cases with little chance of success. It is possibly more likely that differential success rates are the result of authorities taking a 'harder' or a softer' approach in deciding what cases fit the criteria for exceptional hardship

Why is expenditure lower than expected?

What is perhaps more surprising than the variation between authorities in the awards they make and therefore their expenditure, is the apparently very low level of applications compared with the potential number of applications (as indicated by the analysis of the Rent Officer statistics supplied by DSS) It may be the case that, as Chapter 3 suggested, tenants are able to find the shortfall in their rent from elsewhere, or are negotiating lower rents with their landlords. It may also be the case that people are suffering hardship but that there are barriers to applying for an exceptional hardship payment This research, however, was not aimed at understanding non-take-up of exceptional hardship payments Nevertheless, our analysis of some of the application forms and publicity materials collected in the course of the project, suggests strongly that they have the possibility, or even the probability of putting off a large proportion of claimants from applying Many were long and required an immense amount of detail which could be difficult to provide. Detailed information about household expenditure in particular could well discourage a lot of claimants. At this stage, this must remain an hypothesis, although other research on take-up of benefits (such as Corden, 1995 van Oorschot 1995) would indicate that it is a strong hypothesis

The research has indicated that, when official returns are made later in the year, expenditure on exceptional hardship payments for 1997/98 will be found to have increased substantially over the previous year Expenditure is on the increase although the budget for exceptional hardship payments is still considerably under-spent

How much is the average exceptional hardship payment?

Chapter 5 has shown the wide variation in the amounts of money paid by authorities in exceptional hardship payments. Amounts that are small, and for relatively short periods may be justified on the criteria used by an authority, as much as payments that are large and likely to continue for a long time. Most exceptional hardship payments, however, uppear to be at the lower end of the spectrum up to $\chi_2^{(2)}$ per week

The analysis of estimated amounts of money accruing to individual claimants over the durition of an exceptional hardship payment showed that in some authorities claimants were receiving average amounts that were substantial, in excess of $\int 500 \text{ m}$ a quarter of authorities and more than $\int 1000 \text{ m}$ over 20 authorities.

What are local authorities' views about their allocations for exceptional hardship payments from central government²

Chapter 2 showed that while some of the lower spending authorities considered their budgets too high, the majority thought them about right This could be interpreted as a somewhat inconsistent set of responses considering the evidence that most authorities spend only a small fraction of their budgets. The most common explanation for the dominant view was that an increase in applications was expected. This view is not unrealistic since experience tells most Housing Benefit managers that demand for additional payments, from whatever source inexorably increases as awareness of their availability spreads. The evidence of the increased spend in the second year of the scheme further supports their view.

Apart from increased demand, there is another reason why expenditure is likely to increase each year. This was suggested by the finding from Chapter 5 that exceptional hardship payments tend to be awarded either as temporary measures or as ongoing pryments to claimants whose exceptional hardship is expected to continue indefinitely. It is possible to foresee, therefore, a scenario for future trends in exceptional hardship payment expenditure that resembles the pattern of growth for the old Invalidity Benefit (i.e. before its replicement by Incapacity Benefit, a policy change intended to stem the rise in the number of awards) For Invalidity Benefit, the steeply rising cost of awards was explained principally by the growing number of long term recipients rather than by an increase in the caselond Similarly, one can foresce a flow on and off exceptional hardship payments of claimants whose circumstances giving use to exceptional hardship are temporary, but at the same time an mexorable rise in the numbers of 'permanent' recipients whose circumstances are unlikely to change

The long term picture in such a scenario is a growing demand for exceptional hardship payments in each authority year on year until a point when expenditure approaches and reaches the permitted total

4

<u>52</u>

Do particular types of claimants tend to be more or less successful in applying for an exceptional hardship payment than others?

It was difficult for some authorities to provide information about the characteristics of claimants who were either successful or unsuccessful in their application for a payment. However, from the data we have some interesting findings emerged. It appears that some types of claimant had high success rates compared with others. Disabled people, people with mental health problems and pregnant women appeared to have the greatest chance of success. The numbers of applications from people under 25 was comparable to the number of applications from disabled people but they were not as successful.

Lone parents were no more successful than pensioners but applied in far greater numbers and so emerged as the largest group of beneficiaries. The low level of applications from pensioners was striking but perhaps not surprising given their low take-up of benefits in general. However, this finding does indicate that greater targeting of publicity of a more pro-active approach by authorities is needed to overcome the apparent reluctance of pensioners to apply.

How do claimants become aware of the availability of exceptional hardship payments?

Chapter 3 described the various ways in which Housing Benefit claimants are informed about the availability of exceptional hardship payments. Three-quarters of the authorities in the sample included information in the decision letters notifying claimants of the outcome of their benefit claim. Over two-thirds disseminated information through local advice agencies. Leaflets were another common means of publicity.

Other means of publicity, which might be usefully employed in other authorities included, meetings with claimants, welfare groups or landlords, posters, newsletters the local press, or direct mailshots. Most authorities publicised exceptional hardship payments in more than one way, but ten authorities in the sample reported that they did not publicise their availability at all. As mentioned earlier, the survey evidence suggests that the number of different ways in which exceptional hardship payments were publicised influenced the number of applications from claimants. We were not able to assess the quality, style or content of the publicity, but it is likely that the way and the detail in which the scheme is described will serve either to encourage or discourage applications. How do local authorities make decisions on applications for an exceptional hardship payment?

What criteria do they use?

Does a pre-tenancy determination affect the decision whether to make a payment?

Chapter 4 described how most authorities use either the DSS guidance their own policy or set of criteria, or both to help them decide applications for exceptional hardship payments. The extent to which the DSS guidance was useful varied between authorities. Some Housing Benefit managers reported that they experienced problems with the scheme because of a *lack* of guidance on what constitutes 'exceptional hardship'. Other authorities were apparently content to rely solely on the DSS *circular for* guidance

Pre-tenancy determinations were clearly a consideration for most authorities in deciding applications. There was a general feeling that a claimant who had taken on a rented property in the knowledge that they were unlikely to get Housing Benefit to the full amount of the rent would have to present a stronger case than other claimants to establish exceptional hardship'. Some authorities were also suspicious of claimants who did not ask for a pre-tenancy determination before taking a tenancy. They considered that since these had been available for around two years claimants *should* know about them and request one in all cases. If a pretenancy determination was not requested by a claimant, there was a suspicion that this was deliberate in order to be able to claim later that they were unaware that the rent would not be met in full.

Many authorities emphasised that, while they were certainly relevant, pre-tenancy determinations were not the only or main factor when they considered applications

How easy or difficult has it been to implement the exceptional hardship payment scheme?

What is the nature of any problems experienced?

Before the introduction of the exceptional hardship payment scheme in 1996, local authorities were already empowered to make additional payments of Housing Benefit to claimants in exceptional circumstances. One of the main differences with the new scheme is that it is partly funded by central government and has spending ceilings attached. Local authorities are therefore required to introduce systems for managing and controlling the flow of expenditure. Another difference is that awards under the new arrangements must be decided on a test of exceptional hardship rather than the existence of exceptional circumstances. The evidence from the survey suggests that most authorities have managed the introduction of the exceptional hardship payment scheme without mijor difficulties. Difficulties with computer systems were experienced in 43 authorities, but only one authority (in nearly 300) associated their serious' problems with administering exceptional hardship payments with their computer systems

Most of the difficulties with exceptional hardship payments were related either to the intrinsic nature of the scheme (i.e. problems in interpreting what 'exceptional hardship' means), or to day-to-day administration (such as the time and effort required to investigate applications)

Variations between local authorities in the ways they administer aspects Comments on policy development of Housing Benefit policy are the norm rather than the exception. It is not surprising, therefore, to find wide variations in authorities' administration of the exceptional hardship payment scheme. Whether the variations in practice and outcomes are acceptable or not will depend in part on whether one views the right of local authorities to administer discretionary powers in their own way as more or less important than the right of Housing Benefit claimants to be treated with a degree of equity regardless of wherever they happen to live. On the basis of this research, it is fair to infer that applicants and importantly potential applicants, in different local authority areas are not treated equitably. Local variations in rent levels, the number and amounts of rent restrictions, and the demographic constitution of the claimant population will all contribute to differences in application rates, success rates and value of exceptional hardship payments. However it is inconceivible that such variations could explain why some authorities have zero application rates and zero success rates while other authorities attract hundreds of applications and some have 100 per cent success rates

> Clearly there is a need for some form of arrangements that allow claimants in exceptional circumstances or hable to exceptional hardship to obtain help with their rent above the normal entitlement. However, there is some though not conclusive evidence from the study that the low level of applications and awards suggest that the guidance set out in Circular A7/96 could be drawn too tightly such that local authorities making decisions based on them, which they may have amended or added to themselves, are excluding claimants who are suffering a degree of considerable hardship but not to the extent suggested in the guidance. The policy of exceptional hardship payments may, therefore not be helping some of the people it was intended for. One policy response could be to issue further guidance that effectively broadens the number and type of circumstances in which claimants could be helped by the exceptional hardship payment scheme.

> The variations discovered in the documents supplied by local authorities in the course of the research suggest that there is a case for a 'good

practice guide that contains examples of high quality publicity materials appropriate wording to be used in decision letters, application forms and internal documents used in decision making

Ţ.

The policy option of reducing the budget for exceptional hardship payments is always available. The research suggests that the demands on the budget are increasing and likely to continue increasing. Housing Benefit minagers are aware of this also. Any proposed reduction in budgets are therefore likely to generate opposition from a large number of iuthorities

REFERENCES

Corden, A (1995) Changing Perspectives on Benefit Take-up, London HMSO

Sainsbury, R and Eardley, T (1991) Housing Benefit Reviews An evaluation of the effectiveness of the review system in responding to claimants dissatisfied with housing benefit decisions, DSS Research Report No 3 London HMSO

Stafford, B, Kellard, K and Horsley, E (1997) *Customer Contact with the Benefits Agency*, DSS Research Report No 65, London The Stationery Office

van Oorschot, W (1995) Realizing Rights, Aldershot Avebury

APPENDIX I QUESTIONNAIRE FOR TELEPHONE SURVEY

-

—

ł

| Qu | estions 1-13. Quantitative and financial data | a |
|----|---|--|
| 1 | How many claimants were awarded exceptional hardship payments which began between 1 April 1997 and 31 March 1998, that is excluding those that were already in payment before 1 April 1997? | ENTER NUMBER [IF ZERO, GO TO Q 8] |
| 2 | How many of these cases were | [READ OUT AND ENTER NUMBER] lone parents disabled people retired people single under 25s pregnant women people with mental health problems absent parents |
| 3 | How many exceptional hardship payments are currently in payment ² | ENTER NUMBER |
| -4 | How much money was paid out in exceptional hirdship payments in the period 1 April 1997 to 31 March 1998? | ENTER AMOUNT <u>(</u> |
| 5 | Of the payments made in 1997–98 what was the lowest weekly amount ² | ENTER AMOUNT £ per week |
| 6 | And what was the highest weekly amount? | ENTER AMOUNT <u>f</u> per week |
| 7 | And what was the average weekly amount? | ENTER AMOUNT £ per week |
| 8 | How many applications for exceptional hardship payments were turned down in the period 1 April 1997 to 31 March 1998? | ENTER NUMBER |
| y | How many of these cases were | [READ OUT AND ENTER NUMBER] lone parents disabled people retired people single under 25s pregnant women people with mental health problems absent parents |

_

-

-

-

| In 1997-98 how many Housing Benefit awards were restricted on the basis of a local reference rent or a single room rent? | ENTER NUMBER | | |
|--|---|---|--|
| And what proportion made contact with the authority after getting notification of their Housing Benefit award? | [READ OUT CODE ONE RESPONSE ONLY the large majority of cases more than half about half fewer than half only a small minority of cases none at all (DK) | | |
| Do vou think that your 'permitted total' (that is, the total amount of money you are allowed to spend on making exceptional hardship payments) is | Too much Too little About right | ONSE ONLYJ | |
| Why do you say that? | [PROBE AND WRITE IN] | | |
| estions 14-23. The operation of the exception | onal hardship payment scheme | | |
| Is the availability of exceptional hardship payments referred to in | [READ OUT AND ENTER RESI decision letters sent to claimants? Housing Benefit leaflets? Separate form/tear of slip Information supplied to advice agencies? Other answer [WRITE IN] | Ponse] Y n Dk Y n Dk Y n Dk Y n Dk | |
| | awards were restricted on the basis of a local reference rent or a single room rent? And what proportion made contact with the authority after getting notification of their Housing Benefit award? Do vou think that your 'permitted total' (that is, the total amount of money you are allowed to spend on making exceptional hardship payments) is Why do you say that? estions 14-23: The operation of the exception is the availability of exceptional hardship | awards were restricted on the basis of a local reference rent or a single room rent ² And what proportion made contact with the authority after getting notification of their Housing Benefit award ² Do vou think that your 'permutted total' (that is, the total amount of money you are allowed to spend on making exceptional hardship payments) is Stions 14-23: The operation of the exceptional hardship payments referred to in Is the availability of exceptional hardship payments referred to in And what proportion made contact with the authority after getting notification of the set of the availability of exceptional hardship payments referred to in Is the availability of exceptional hardship payments referred to in And what proportion made contact with the authority after getting notification of the sceptional hardship payments referred to in Is the availability of exceptional hardship payments referred to in Authority after getting notification of the information supplied to advice agencies ² | |

| | We are interested in the types of reasons claimants put forward in their applications for exceptional hardship payments. I will read out a number of examples which have been given to us. Can you say in each case whether, in your authority, you have received applications citing these reasons. And could you say whether they are common or not? | [READ OUT AND ENTER RESPONSE] (KEY Y/C = yes a common reason Y/N = yes, but not a common reason N = no examples of this reason in this authority) Claimant has a medical condition or illness Y/C Y/N N DK Claimant has a disability Y/C Y/N N DK Claimant can't afford rent Y/C Y/N N DK Claimant is pregnant Y/C Y/N N DK Claimant has language difficulties Y/C Y/N N DK Claimant can't afford to pay for food Y/C Y/N N DK Claimant can't afford to pay other bills Y/C Y/N N DK Claimant is under 25 Y/C Y/N N DK Claimant needs room for children who stav occasionally Y/C Y/N N DK Any other common reasons ² [WRITE IN] |
|----|---|--|
| | (Routing instruction: IF Q | 2.8 = ZERO, GO TO Q.18) |
| 16 | You said that, in 1997–98 there were [RETRIEVE ANSWER TO Q 8] unsuccessful applications for an exceptional hardship payment How many of these appealed? | [PROBE FOR WHETHER ANSWER IS EXACT OR ESTIMATE] ENTER EXACT NUMBER OR ENTER ESTIMATE (DK) |
| 17 | And how many of these were successful? | [PROBE FOR WHETHER ANSWER IS EXACT OR ESTIMATE] ENTER EXACT NUMBER OR ENTER ESTIMATE (DK) |
| 18 | We are interested in how you are making decisions on exceptional hardship payment applications Which of the following statements describes the situation in your authority? You can answer yes to more than one | [READ OUT AND ENTER RESPONSE] We have our own written policy about what constitutes exceptional hardship Y N DK We have developed a check list of criteria that we use in deciding cases Y N DK We mamly use the DSS guidance Y N DK circular We decide each case on its particular inerits Y N DK [NOW GO TO Q 20] |

ł

_

| 19 | We are interested in how you will make decisions on exceptional hardship payment applications in the future. Which of the following statements describes the situation in your authority? You can answer yes to more than one | [READ OUT AND ENTER RESPONSE]We have our own written policy aboutwhat constitutes exceptional hardship Y N DKWe have developed a check list of criteriathat we will use in deciding cases Y N DKWe will mainly use the DSSguidance circularY N DKWe have not yet developed a policy ondealing with applications for exceptionalhardship payments but plan toY N DKWe decide each case on itsparticular meritsY N DK[NOW GO TO Q 21] |
|----|---|---|
| 20 | Is a Pre-Tenancy Determination a relevant consideration in deciding an exceptional hardship payment application? | Yes No DK [NOW GO TO Q 22] |
| 21 | Will a Pre-Tenancy Determination be a relevant consideration in the future in deciding an exceptional hardship payment application? | Yes No DK |
| 22 | Thinking about what you have done in your authority to enable you to put the exceptional hardship payment scheme into operation – such as changes to administrative or computer systems training etc – how easy or difficult has it been to administer the scheme since January 1996 ² | [RFAD OUT, CODE ONE RESPONSE ONLY] Verv easy [END INTERVIEW] Mainly easy [END INTERVIEW] Some difficulties [GO TO Q 23] Serious problems [GO TO Q 23] |
| 23 | Can you explain what the main problems or difficulties have been? | [WRITE IN] |

T

APPENDIX 2 INTRODUCTORY LETTERS TO LOCAL AUTHORITIES FROM DSS AND SPRU

1

230

_



DEPARTMENT OF SOCIAL SECURITY The Adelphi, 1-11 John Adam Street, London WC2N 6HT Telephone 0171-962 8000

Gin 391

I April 1998

Dear Housing Benefit Manager

Research into Local Authorities' Use of Key Housing Benefit and Council Tax Benefit Discretionary Powers

I am writing to ask for your help with a survey which the Department of Social Security has commissioned on the use of discretionary powers in the administration of Housing Benefit and Council Tax Benefit claims

The aim of the research is to gather information on the use of certain discretionary powers the exceptional hardship payment scheme and the power to withhold benefit where there are doubts about the propriety of the landlord or suspend benefit because of doubts about the eligibility of the claimant. Local Authority Associations have been informed of this research.

We have commissioned the Social Policy Research Unit (SPRU) at the University of York and Public Attitude Surveys Ltd (PAS) to undertake the research on our behalf. The study will be based on telephone interviews with all local authorities. Development work has already been carried out in a small number of local authorities.

We would very much like you to participate in this survey. This will involve a 20 minute telephone interview with the person in your local authority who would be best able to answer questions on exceptional hardship payments and the new discretionary powers. In order to carry out the research within our timetable we need to conduct the interviews after the Easter holiday. The accompanying letter from the Social Policy Research Unit explains more about how the research will be carried out.

Although your participation in this survey is voluntary, it is very important that as many local authorities are involved as possible. I can assure you that all information provided during the research will be treated in strict confidence by the research team. The results of this study will be presented in such a way that no individual or local authority can be identified in the report which is provided to the Department Each participating authority will receive feedback on the findings of the study once it has been completed

They will contact you in the next few days to discuss the research with you If you would like to know more about the study, or if you have any queries, please do not hesitate to get in touch with Rachel Trott (0171 962 8555) here in the Department or the lead researcher from SPRU, Dr Roy Sainsbury (01904 433608)

Thank you in advance for your help with this research

Yours sincerely

Reand Mitton

Bernard Mitton Housing Benefit Policy

1 April 1998

Dear Housing Benefit Manager

Research into Local Authorities' Use of Key Housing Benefit and Council Tax Benefit Discretionary Powers

Thank you for agreeing to take part in this project. The accompanying letter and fact sheet from the DSS explain the scope of the study and the roles played by the research team at SPRU and the survey agency PAS Ltd. I am sure that you will find the findings from the research interesting and useful

I have enclosed a 'questionnaire outline' which explains the questions we would like to ask you during the interview. The work of developing the questionnaire has been assisted by the staff of 18 authorities in Great Britain who were visited by members of the York research team in February and March of this year.

Some of the questions require statistical data relating to your discretionary powers and you may have to prepare these in advance of the actual interview. The outline is intended as a guide. The exact wording of the questions may change following pilot testing of the questionnaire this week.

The discretionary powers that are the subject of the project fall into two distinct types. The first concern your powers established initially in January 1996 to grant exceptional hardship payments to claimants whose Housing Benefit awards have been restricted on the basis of a local reference rent or single room rent (Circular HB/CTB A7/96 contains the details). We ask about these in Section A of the questionnaire

The second set of powers derive from the Social Security Administration (Fraud) Act 1997 and are intended to increase the capability of local authorities to combat Housing Benefit fraud. These are set out in Circular HB/CTB A48/97. We are aware that these powers only came into force in November 1997 but the Department is very interested in authorities' early experiences (or plans for the future). We ask about the new powers which relate to landlords in Section B of the questionnaire, and those relating to claimants in Section C.

The next step in the research study is that PAS will contact you direct to book a convenient time after the Easter holiday for the interview to take place. If you have any queries regarding the research I will be happy to answer them as best I can

Yours sincerely

Dr Roy Sainsbury Research Team Leader

OTHER RESEARCH REPORTS AVAILABLE:

_

| No | Title | ISBN | Price |
|----|---|---------------------|--------------------|
| 1 | Thirty Families Their living standards in unemployment | 0 11 761683 4 | £6 65 |
| 2 | Disability, Household Income & Expenditure | 0 11 761755 5 | £5 65 |
| 3 | Housing Benefit Reviews | 0 11 761821 7 | £16 50 |
| 4 | Social Security & Community Care The case of the Invalid Care Allowan | 0 11 761820 9 ce | £9 70 |
| 5 | The Attendance Allowance Medical Examination Monitoring consumer views | 0 11 761819 5 | £5 50 |
| 6 | Lone Parent Families in the UK | 0 11 761868 3 | £15 00 |
| 7 | Incomes In and Out of Work | 0 11 761910 8 | £17 20 |
| 8 | Working the Social Fund | 0 11 761952 3 | £9 00 |
| 9 | Evaluating the Social Fund | 0 11 761953 1 | $\mathcal{L}22 00$ |
| 10 | Benefits Agency National Customer Survey 1991 | 0 11 761956 6 | £16 00 |
| 11 | Customer Perceptions of Resettlement Units | 0 11 761976 6 | £13 75 |
| 12 | Survey of Admissions to London Resettlement Units | 0 11 761977 9 | £8 00 |
| 13 | Researching the Disability Working Allowance Self Assessment Form | 0 11 761834 9 | £7 25 |
| 14 | Child Support Unit National Client Survey 1992 | 0 11 762060 2 | £15 00 |
| 15 | Preparing for Council Tax Benefit | 0 11 762061 0 | £5 65 |
| 16 | Contributions Agency Customer Satisfaction Survey 1992 | 0 11 762064 5 | £18 00 |
| 17 | Employers' Choice of Pension Schemes Report of a qualitative study | 0 11 762073 4 | £5 00 |
| 18 | GPs and IVB A qualitative study of the role of GPs in the award of Invalidity Benefit | 0 11 762077 7 | £12 00 |
| 19 | Invalidity Benefit A survey of recipients | 0 11 762087 4 | £10 75 |

<u>69</u>

I

| 20 | Invalidity Benefit A longitudinal survey of new recipients | 0 11 762088 2 | £19 95 |
|----|--|----------------|-----------------|
| 21 | Support for Children A comparison of arrangements in fifteen countries | 0 11 762089 0 | <u>Д</u> 22 95 |
| 22 | Pension Choices A survey on personal pensions in comparison with other pension options | 0 11 762091 2 | £18 95 |
| 23 | Crossing National Frontiers | 0 11 762131 5 | £17 75 |
| 24 | Statutory Sick Pay | 0 11 762147 1 | L23 75 |
| 25 | Lone Parents and Work | 0 11 762147 X | £12 95 |
| 26 | The Effects of Benefit on Housing Decisions | 0 11 762157 9 | £18 50 |
| 27 | Making a Claim for Disability Benefits | 0 11 762162 5 | £12 95 |
| 28 | Contributions Agency Customer Satisfaction Survey 1993 | 0 11 762220 6 | £2 0 00 |
| 29 | Child Support Agency National Client Satisfaction Survey 1993 | 0 11 762224 9 | L33 00 |
| 30 | Lone Mothers | 0 11 762228 1 | £16 75 |
| 31 | Educating Employers | 0 11 762249 4 | L8 50 |
| 32 | Employers and Family Credit | 0 11 762272 9 | L13 50 |
| 33 | Direct Payments from Income Support | 0 11 762290 7 | £16 50 |
| 34 | Incomes and Living Standards of Older People | 0 11 762299 0 | £24 95 |
| 35 | Choosing Advice on Benefits | 0 11 762316 4 | £13 95 |
| 36 | First-time Customers | 0 11 762317 2 | £25 00 |
| 37 | Contributions Agency National Chent Satisfaction Survey 1994 | 0 11 762339 3 | <u> </u> |
| 38 | Managing Money in Later Life | 0 11 762340 7 | £22 ()() |
| 39 | Child Support Agency National Chent Satisfaction Survey 1994 | 0 11 762341 5 | £35 00 |
| 40 | Changes in Lone Parenthood | 0 11 7632349 0 | £20.00 |
| 41 | Evaluation of Disability Living Allowance and Attendance Allowance | 0 11 762351 2 | £40.00 |
| 42 | War Pensions Agency Customer Satisfiction Survey 1994 | 0 11 762358 X | £18.00 |
| 43 | Paving for Rented Housing | 0 11 762370 9 | £19.00 |

÷

<u>70</u>

t

| 44 | Resettlement Agency Customer | 0 11 762371 7 | £16 00 |
|----|--|---------------|---------------------|
| 45 | Satisfaction Survey 1994 Changing Lives and the Role of | 0 11 762405 5 | £20 00 |
| | Income Support | | |
| 46 | Social Assistance in OECD Countries Sviithesis Report | 0 11 762407 1 | £22.00 |
| 47 | Social Assistance in OECD Countries Country Report | 0 11 762408 X | £47 00 |
| 48 | Leaving Family Credit | 0 11 762411 X | $\mathcal{L}18$ 00 |
| 49 | Women and Pensions | 0 11 762422 5 | £35 00 |
| 50 | Pensions and Divorce | 0 11 762423 5 | $\mathcal{L}25 00 $ |
| 51 | Child Support Agency Client Satisfaction Survey 1995 | 0 11 762424 1 | £22 00 |
| 52 | Take Up of Second Adult Rebate | 0 11 762390 3 | £17-00 |
| 53 | Moving off Income Support | 0 11 762394 6 | £26 00 |
| 54 | Disability, Benefits and Employment | 0 11 762398 9 | £30-00 |
| 55 | Housing Benefit and Service Charges | 0 11 762399 7 | £25 00 |
| 56 | Confidentiality The public view | 0 11 762434 9 | £25 00 |
| 57 | Helping Disabled Workers | 0 11 762440 3 | £25 00 |
| 58 | Employers' Pension Provision 1994 | 0 11 762443 8 | £30-00 |
| 59 | Delivering Social Security A cross- national study | 0 11 762447 0 | £35 00 |
| 60 | A Comparative Study of Housing Allowances | 0 11 762448 9 | £26 00 |
| 61 | Lone Parents, Work and Benefits | 0 11 762450 0 | £25 00 |
| 62 | Unemployment and Jobseeking | 0 11 762452 7 | £3 0 00 |
| 63 | Exploring Customer Satisfaction | 0 11 762468 3 | £20 00 |
| 64 | Social Security Fraud The role of penalties | 0 11 762471 3 | £30 00 |
| 65 | Customer Contact with the Benefits Agency | 0 11 762533 7 | £30 00 |
| 66 | Pension Scheme Inquiries and Disputes | 0 11 762534 5 | £30.00 |
| 67 | Maternity Rights and Benefits in Britain | 0 11 762536 1 | £35 00 |
| 68 | Claimants' Perceptions of the Claim Process | 0 11 762541 8 | £23 00 |
| 69 | Dehvering Benefits to Unemployed People | 0 11 762553 1 | £27 00 |

، ۱

<u>71</u>

| 70 | Delivering Benefits to Unemployed 16–17 year olds | 0 11 762557 4 | £20.00 |
|----|---|-----------------------|------------------------|
| 71 | Stepping–Stones to Employment | 0 11 762568 X | £27 00 |
| 72 | Dynamics of Retirement | 0 11 762571 X | £36.00 |
| 73 | Unemployment and Jobsceking before Jobseeker's Allowance | 0 11 762576 0 | £34 00 |
| 74 | Customer views on Service Delivery in the Child Support Agency | 0 11 762583 3 | £27 00 |
| 75 | Experiences of Occupational Pension Scheme Wind–Up | 0 11 762584 1 | £27 00 |
| 76 | Recruiting Long-Term Unemployed People | 0 11 762585 X | L27 ()() |
| 77 | What Happens to Lone Parents | 0 11 762598 3 | \mathcal{L}^{31} 00 |
| 78 | Lone Parents Lives | 0 11 762598 1 | L34 00 |
| 79 | Moving into Work Bridging Housing Costs | 0 11 762599 X | £33 00 |
| 80 | Lone Parents on the Margins of Work | 1 84123 000 6 | £26 00 |
| 81 | The Role of Pension Scheme Trustees | 1 84123 001 4 | $\mathcal{L}^{28 00}$ |
| 82 | Pension Scheme Investment Policies | 1 84123 002 2 | $\mathcal{L}^{28 00}$ |
| 83 | Pensions and Retirement Planning | 1 84123 003 0 | \mathcal{L}^{28} 00 |
| 84 | Self–Employed People and National Insurance Contributions | 1 84123 004 9 | £28 00 |
| 85 | Getting the Message Across | 1 84123 052 9 | $\mathcal{L}^{26 00 }$ |
| 86 | Leaving Incapacity Benefit | 1 84123 087 1 | £34 (1() |
| 87 | Unemployment and Jobseeking Two Yeats On | 1 84123 088 X | £38 00 |
| 88 | Attitudes to the Welfare State and the Response to Reform | 1 84123 098 7 | £36 00 |
| 89 | New Deal for Lone Parents Evaluation of Innovative Schemes | 1 84 123 101 0 | £26 00 |
| 90 | Modernising service deliverv The Lone Parent Prototype | 1 84123 103 7 | £26 00 |
| | Social Security Research Yearbook 1990–91 | 0 11 761747 4 | £8 00 |
| | Social Security Research Yearbook 1991–92 | 0 11 761833 0 | £12 00 |
| | Social Security Research Yearbook 1992–93 | 0 11 762150 1 | £13 75 |

í

4

<u>11</u>

| Social Security Research Yearbook 1993–94 | 0 11 762302 4 | £16 50 |
|--|----------------|--------|
| Social Security Research Yearbook 1994–95 | 0 11 762362 8 | £20 00 |
| Social Security Research Yearbook 1995–96 | 0 11 761446 2 | £20 00 |
| Social Security Research Yearbook 1996–97 | 0 11 762570 1 | £27 00 |
| Social Security Research Yearbook 1997–98 | 1 84 123 086 3 | L34 00 |

Further information regarding the content of the above may be obtained from

Department of Social Security Attn Keith Watson Social Research Branch Analytical Services Division 5 4th Floor, Adelphi 1–11 John Adam Street London WC2N 6HT Telephone 0171 962 8557 local authorities have a range of disactionery powers which can be used in the administration of Housing Bandit. These induces the power to make exceptional heredship payments to definents whose Housing Bandit is less than their cant because of regulations introduced since 1996. These regulations incorporated the local reflectors cant and the single coom cant into Housing Bandit adhering are based on telephone interviews with Housing Bandit managers. The Social Policy Research Unit at the University of York canted out the research.

The Dependment of Social Security Social Research Branch is responsible for commissioning and managing the Dependments research programme. The research programme serves the information needs of Ministers, the Dependment and its Agandes. Research contributes to the development and implementation of new policies, and the monitoring and evaluation of existing policies. It also plays an important role in providing customer faceback on the Dependment's services.

Researchers in the branch work descrip with Departmental colleagues responsible for making policy and delivering the services for which the Department is responsible. Most projects are cauted out by external contractors who work dosely with in-house researchers throughout the project. A programme of work is agreed annually and details are announced.

If you would like to know more about the work of the Social Research Branch plasse contacts

Reith Watson, Social Rescerch Branch, Ath Floor Addphi, 1-01 John Adam Street, London WQN AHL E Keill - keith@astlandandss-astlgounk Internat address - http://www.dss.gov.nl/astl/astl5/



Corporate Document Services



 \mathcal{A}